#### ARTICLE

# Politics and Justice at the International Criminal Court

#### Richard H Steinberg

Jonathan D Varat Endowed Professor of Law, Professor of Political Science (by courtesy), University of California, Los Angeles (United States) Email: steinberg@law.ucla.edu

#### Abstract

The International Criminal Court (ICC) is a legal institution embedded in international politics. Politics shaped the Rome Statute of the ICC, which is rooted in norms and rules of European lineage and security interests of party states. Politics constrains and influences the operation of the Court, which has adapted in response to oversight and governance of the Assembly of States Parties, and to political actions extrinsic to institutional rules. The ICC also has political effects in situation states. A brief history shows that application of Rome Statute triggers across state parties with different social conditions skewed geographic distribution of its investigations and prosecutions towards Africa, a structural bias that catalysed a legitimation crisis for the ICC. Subsequent exercises of expansive jurisdiction aimed at nationals of non-African, non-party states - including Israel and some of the world's great powers - have dampened African complaints and advanced the ICC agenda, but intensified non-legitimacy claims by powerful non-party states. To survive, Court organs must follow legal mandates, yet be responsive to pressing international political demands, continuously risking the legitimacy of the ICC as a legal institution and adverse political reactions by antagonised governments. Careful management of the tension between law and politics at the ICC may modestly reduce antagonism towards the Court, but that tension cannot be resolved, and confrontations over the ICC's legitimacy are certain to recur.

Keywords: International Criminal Court; ICC; politics; justice; Rome Statute

# I. Introduction

When the Rome Statute entered into force, constituting the International Criminal Court (ICC), many were hopeful that the Court might dampen

© The Author(s), 2024. Published by Cambridge University Press in association with the Faculty of Law, the Hebrew University of Jerusalem. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.

violence and atrocities in the world. The ICC offered a promise of enhanced accountability for perpetrators of atrocities, justice for survivors, deterrence of atrocities, and removal of perpetrators from situations where they could commit more atrocities. Many expected that the ICC might achieve all this as a pure legal institution, applying law to facts, treating similar cases similarly, and speaking law to power – behaviour that creates a sense that justice is being done and lends law legitimacy in domestic and international legal systems.<sup>1</sup>

So, it should not be surprising that the Court's mantra has been exactly that: take decisions for legal reasons exclusively. The first Prosecutor, Luis Moreno Ocampo, repeatedly asserted that politics would not enter into his decisions, stating at the outset: 'I shall not be involved in political considerations. I have to respect scrupulously my legal limits'.<sup>2</sup> The second Prosecutor, Fatou Bensouda, stated her policy: 'We will do so with unyielding commitment to end impunity for mass crimes and in total independence, but we can only do so in strict conformity with the Rome Statute legal framework'.<sup>3</sup> The third and current Prosecutor, Karim Khan, declared early in his tenure: 'As I proceed to discharge my responsibilities, I will ensure that investigations by my office are conducted objectively and independently'.<sup>4</sup> Moreover, the ICC has all the hallmarks of a legal institution: a statute elaborately defining the crimes addressed;<sup>5</sup> strict rules of procedure and evidence;<sup>6</sup> specified rights of the accused;<sup>7</sup> a hierarchy of Chambers (Pre-trial, Trial, and Appeals);<sup>8</sup> the

<sup>&</sup>lt;sup>1</sup> 'Legitimacy', as used in this article, is conceived as social legitimacy, defined as the property projected onto an action, rule, or system by an actor's belief that the action, rule, or system is morally or legally legitimate; it may be assessed empirically. See Christopher A Thomas, 'The Uses and Abuses of Legitimacy in International Law' (2014) 34 *Oxford Journal of Legal Studies* 729. See also Ian Hurd, *After Anarchy: Legitimacy and Power in the United Nations Security Council* (Princeton University Press 2007). See generally Max Weber, 'Bureaucracy' in Hans H Gerth and C Wright Mills (eds), *Max Weber: Essays in Sociology* (Routledge 2014) 196; Allen Buchanan, *Justice, Legitimacy, and Self-Determination: Moral Foundations of International Law* (Oxford University Press 2003); Thomas M Franck, *The Power of Legitimacy Among Nations* (Oxford University Press 1990).

<sup>&</sup>lt;sup>2</sup> Luis Moreno-Ocampo, 'Keynote Address', speech delivered at the Council on Foreign Relations, Washington DC (United States), 4 February 2010, 6, https://www.icc-cpi.int/sites/default/files/NR/rdonlyres/A80CDDDD-8A9A-432E-97CE-F6EAD700B5AE/281527/100204ProsecutorsspeechforCFR.pdf.

<sup>&</sup>lt;sup>3</sup> Fatou Bensouda, 'Statement of the Prosecutor of the International Criminal Court, Fatou Bensouda, on Concluding the Preliminary Examination of the Situation Referred by the Union of Comoros: "Rome Statute Legal Requirements Have Not Been Met", transcription of audio-visual statement, 6 November 2014, https://www.icc-cpi.int/news/statement-prosecutor-international-criminal-court-fatou-bensouda-concluding-preliminary.

<sup>&</sup>lt;sup>4</sup> Karim Khan, 'Statement of ICC Prosecutor, Karim A.A. Khan QC, on the Situation in Ukraine: Receipt of Referrals from 39 States Parties and the Opening of an Investigation', 2 March 2022, https://www.icc-cpi.int/news/statement-icc-prosecutor-karim-aa-khan-qc-situation-ukrainereceipt-referrals-39-states.

 $<sup>^5</sup>$  Rome Statute of the International Criminal Court (entered into force 1 July 2002) 2187 UNTS 90 (Rome Statute).

<sup>&</sup>lt;sup>6</sup> ICC Rules of Procedure and Evidence, Official Records of the ASP to the Rome Statute of the ICC, 1st session, New York, 3–10 September 2002, ICC-ASP/1/3 and Corr.1, Pt II.A. (ICC RPE).

<sup>&</sup>lt;sup>7</sup> See Rome Statute (n 5) Pt 6. See also, eg, ICC RPE (n 6) rr 20, 27.

<sup>&</sup>lt;sup>8</sup> Rome Statute (n 5) arts 34, 39.

decorum and pomp of a courtroom; refined modes of argumentation; and expansive case law.  $\!\!\!^9$ 

However, like all international organisations, the ICC is nested in global politics. International criminal law has been deeply affected by politics from the beginning. The trial of Conradin Von Hohenstaufen in 1268 is said to be the first post-classical European trial for war crimes.<sup>10</sup> Conradin, who carried the title of King of Jerusalem, was tried purportedly for murder and pillaging in Tagliacozzo, a town outside Naples, found guilty of a crime 'against the laws of God and man', and was executed. However, historians who have studied the events argue that this was actually a political trial and execution, directed by the Pope, who preferred that a French noble control the Naples region.<sup>11</sup>

Centuries later, the parties to the Treaty of Versailles indicted Kaiser Wilhelm II for 'a supreme offence of crimes against international morality and the sanctity of treaties' and specified that a tribunal was to be constituted to try the accused;<sup>12</sup> others were also to be tried for various war crimes.<sup>13</sup> However, the special tribunal for Wilhelm was never constituted and he was never tried; the Kaiser was exiled to the Netherlands; the British did not consistently pressure the Dutch to extradite him; and the United States government feared German unrest if he were extradited and tried.<sup>14</sup> Politics again interfered with justice.

After the Second World War, Emperor Hirohito of Japan was not charged by the International Military Tribunal for the Far East,<sup>15</sup> despite his having done nothing to stop the Rape of Nanking where 200,000 people

<sup>13</sup> ibid arts 228-30.

<sup>14</sup> Nigel J Ashton and Duco Hellema, 'Hanging the Kaiser: Anglo-Dutch Relations and the Fate of Wilhelm II, 1918–20' (2007) 11(2) *Diplomacy & Statecraft* 53, 63, 70. See also Arthur Waworth, *Wilson and His Peacemakers: American Diplomacy at the Paris Peace Conference* (WW Norton and Co 1986).

<sup>15</sup> Charter of the International Military Tribunal for the Far East (entered into force 19 January 1946, amended 26 April 1946) TIAS 1589.

<sup>&</sup>lt;sup>9</sup> See International Criminal Court, online database, https://legal-tools.org/cld.

<sup>&</sup>lt;sup>10</sup> In classical Greece, Lysander, the Spartan commander, convened his allies in 405 BC to decide the fate of captured Athenians who had allegedly committed illegal acts in wartime, and decided to throw most into the sea to their deaths. Some have referred to this as a tribunal: Timothy LH McCormack, 'From Sun Tzu to the Sixth Committee: The Evolution of an International Criminal Law Regime' in Timothy LH McCormack and Gerry J Simpson (eds), *The Law of War Crimes: National and International Approaches* (Kluwer 1997) 31

<sup>&</sup>lt;sup>11</sup> M Cherif Bassiouni, 'Perspectives on International Criminal Justice' (2010) 50 Virginia Journal of International Law 269, 297.

<sup>&</sup>lt;sup>12</sup> Article 227 of the Treaty of Versailles provides: 'The Allied and Associated Powers publicly arraign William II of Hohenzollern, formerly German Emperor, for a supreme offence against international morality and the sanctity of treaties. A special tribunal will be constituted to try the accused, thereby assuring him the guarantees essential to the right of defence. It will be composed of five judges, one appointed by each of the following Powers: namely, the United States of America, Great Britain, France, Italy and Japan. In its decision the tribunal will be guided by the highest motives of international policy, with a view to vindicating the solemn obligations of international undertakings and the validity of international morality. It will be its duty to fix the punishment which it considers should be imposed. The Allied and Associated Powers will address a request to the Government of the Netherlands for the surrender to them of the ex-Emperor in order that he may be put on trial': Treaty of Versailles (entered into force 10 January 1920) art 227.

were killed,<sup>16</sup> and nothing to stop the sexual slavery of 300,000 'comfort women'.<sup>17</sup> The political reason for not trying the Emperor is made clear in a 1946 message to Washington from General Douglas MacArthur, arguing that Hirohito's indictment and execution could trigger a 'tremendous convulsion among the Japanese people' and 'a condition of underground chaos and disorder amounting to guerilla warfare'.<sup>18</sup> MacArthur estimated that if Hirohito were tried, one million occupation troops might be needed and 'all hope of introducing democratic methods would disappear'.<sup>19</sup>

In Europe, the Nuremburg trials<sup>20</sup> were also skewed by politics. Soviet forces had perpetrated mass atrocities, including mass rape and murder, as they drove Nazi forces back into Germany.<sup>21</sup> When the Nuremberg Charter, which established the Tribunal, was being negotiated, representatives of the United States and the United Kingdom pressed for some Soviets to be tried. Not surprisingly, Stalin objected and threatened to end all Allied cooperation in Europe; no Soviets were tried.<sup>22</sup>

All this suggests that in the international context, Lady Justice might not be blind. There is no shortage of arguments that international justice is merely 'victor's justice',<sup>23</sup> and most of the examples above of earlier international justice efforts may be seen in that light, with powerful states championing and establishing international justice mechanisms that selectively prosecute military and civilian leaders of weaker states whose military has usually lost a conflict. In so far as the ICC has been perceived as a European-led effort which has mainly investigated and prosecuted people from former European colonies in Africa,<sup>24</sup> it too may be seen as analogous to victor's justice, to the extent that the powerful constituted a court that is prosecuting the weak.

Another posited tension between law and politics at the ICC suggests that, at least in some cases, justice through international criminal tribunals such as the

<sup>19</sup> ibid.

<sup>&</sup>lt;sup>16</sup> Iris Chang, The Rape of Nanking: The Forgotten Holocaust of World War II (Basic Books 1997).

<sup>&</sup>lt;sup>17</sup> George Hicks, The Comfort Women: Japan's Brutal Regime of Enforced Prostitution in the Second World War (WW Norton & Company 1995).

<sup>&</sup>lt;sup>18</sup> Douglas MacArthur, 'General of the Army Douglas MacArthur to the Chief of Staff, United States Army (Eisenhower)', *Foreign Relations of the United States, 1946, The Far East, Vol VIII* transcriptions of telegrams, Document 308 (894.001 Hirohito/1–2546), 396, https://history.state.gov/historicaldocuments/frus1946v08/d308,

 $<sup>^{20}</sup>$  Agreement for the Prosecution and Punishment of the Major War Criminals of the European Axis and Charter of the International Military Tribunal (entered into force 8 August 1945) 82 UNTS 279.

<sup>&</sup>lt;sup>21</sup> Antony Beevor, Berlin: The Downfall 1945 (Viking 2002) 326-27.

<sup>&</sup>lt;sup>22</sup> Francine Hirsch, Soviet Judgment at Nuremberg: A New History of the International Military Tribunal after World War II (Oxford University Press 2020).

<sup>&</sup>lt;sup>23</sup> See, eg, Richard Minear, Victor's Justice: The Tokyo War Crimes Trial (Princeton University Press 1971); David Irving, Nuremberg: The Last Battle (Focal Point 1996).

<sup>&</sup>lt;sup>24</sup> eg, Kamari Maxine Clarke, 'Why Africa?' in Richard H Steinberg (ed), *Contemporary Issues Facing the International Criminal Court* (Brill 2016) 326; Kamari M Clarke, Abel S Knottnerus and Eefje de Volder (eds), *Africa and the ICC Perceptions of Justice* (Cambridge University Press 2016); Charles Achaleke Taku, 'International Politics and Policy Considerations for the Inappropriate Targeting of Africa by the ICC OTP' in Steinberg, ibid 338.

ICC might be more likely to deter peace than deter atrocities: once a leader has perpetrated atrocities and faces the threat of prosecution, the argument goes, that leader is less likely than otherwise to conclude a peace agreement, which would risk them ending up in the dock. They are therefore more likely to fight to the end, with a view to a victory that would surround them with a guard that can protect against their arrest.<sup>25</sup> A contrary position is that the threat of justice, accountability, and prison will deter atrocities.<sup>26</sup> A third view, with considerable statistical evidence behind it, is that both phenomena have been taking place during the ICC era: civilian killing by state party governments is being deterred and diminished in civil conflicts, but civil conflicts are lasting longer, presumably after a leader has engaged in atrocities.<sup>27</sup>

This article focuses on the broader question of how ICC organs and party states have navigated the tension between law and politics. As a legal institution embedded in global politics, the ICC was born with a congenital contradiction. It derives legitimacy by doing what is appropriate under Rome Statute principles and rules, pursuing justice according to the rule of law, applying the law as written to facts to determine the legality of an act or omission. However, legal action pursuant to the Rome Statute often challenges the interests of powerful political actors, and may appear illegitimate to targeted leaders and their followers, especially to those in non-party states subjected to the Court's jurisdiction. To survive, the ICC must often act politically, influenced by pressure from powerful actors to exercise its authority in ways they favour. Wedged between the law of the Rome Statute and global politics, a logic of appropriateness and a logic of consequences,<sup>28</sup> the ICC faces challenges from which it cannot completely escape.

This antinomy is illustrated here historically. Ideationally, the establishment of the ICC became a fundamentally European project, driven by norms, reflected in detailed substantive rules, procedures and practices, which had been developed over centuries in Europe and accepted widely there. Those norms and rules were accepted by many elites in western-influenced former colonies in the global south. The world's great powers – the United States, Russia, China, and India – are not state parties, as a result of a combination of normative differences and security interests.

During the ICC's first decade of operation, a legally conservative application of Rome Statute triggers, in which rules and their application were not

<sup>&</sup>lt;sup>25</sup> Jack Goldsmith and Stephen D Krasner, 'The Limits of Idealism' (2003) 132 Daedalus 47.

<sup>&</sup>lt;sup>26</sup> Ralph Henham, 'The Philosophical Foundations of International Sentencing' (2003) 1 *Journal of International Criminal Justice* 64; Mark B Harmon and Fergal Gaynor, 'Ordinary Sentences for Extraordinary Crimes' (2007) 5 *Journal of International Criminal Justice* 683; Kathryn Sikkink, *The Justice Cascade: How Human Rights Prosecutions Are Changing World Politics* (WW Norton 2011).

<sup>&</sup>lt;sup>27</sup> Daniel Krcmaric, *The Justice Dilemma: Leaders and Exile in an Era of Accountability* (Cornell University Press 2020). See also Hyeran Jo and Beth Simmons, 'Can the International Criminal Court Deter Atrocity?' (2016) 70 International Organization 443.

<sup>&</sup>lt;sup>28</sup> On the logic of appropriateness versus the logic of consequences see James G March and Johan P Olsen, 'The Institutional Dynamics of International Political Orders' (1998) 52 *International Organization* 943.

stretched,<sup>29</sup> resulted in investigations and prosecutions of only Africans. That is not surprising, as Rome Statute crimes are more likely to take place in poor, weak or failed party states than in developed European party states with established borders and strong human rights norms.<sup>30</sup> Moreover, while the United Nations (UN) Security Council may refer cases to the ICC,<sup>31</sup> the permanent members have blocked or vetoed Security Council efforts to refer situations that could result in trying their nationals or those of allies. Atrocities at least as grave as those in Africa were being perpetrated in non-party states like Syria, Iraq, Myanmar, and China, where the terms of the Rome Statute and powerful UN Security Council permanent members made it difficult to assert ICC jurisdiction.

While there is no consensus on what is 'justice', which is an essentially contested concept,<sup>32</sup> many intuitively accept arguments based on the 'like cases' maxim – that similar cases should be treated similarly, and dissimilar cases dissimilarly.<sup>33</sup> Hence, many have seen the overall pattern of ICC investigations and prosecutions as unjust.<sup>34</sup> An African political reaction – based on arguments that Africa is being treated differently from the global north, and fuelled by prosecution of powerful African leaders – challenged the ICC's legitimacy, generated African claims of European neo-colonialism, and resulted in many African states ceasing to cooperate with the Court; some even threatened to withdraw from the ICC, thus imperilling its survival.

Starting in 2011, the ICC adapted with the election of two successive prosecutors generally favoured by African party states, and expansive assertions of jurisdiction that stretched its legal authority to investigate nationals of nonparty states, such as those of the United States in Afghanistan, Myanmar in Bangladesh, and Israel in 'the State of Palestine'. The Court also launched investigations of Russia in Georgia and Ukraine. Those events modestly dampened African complaints, but rekindled another legitimacy challenge – this one from non-party states that found their leaders in the crosshairs of a treaty regime they have not joined, with definitions of crimes or procedures they do

 $<sup>^{29}</sup>$  Moreno-Ocampo (n 2) 6 ('Our policy is never to stretch the interpretation of the norms adopted in Rome. This is the only way to build a judicial institution').

<sup>&</sup>lt;sup>30</sup> It is well established that democracy and human rights are correlated with levels of economic development and associated social and institutional structures; see Seymour Martin Lipset, 'Some Social Requisites of Democracy: Economic Development and Political Legitimacy' (1959) 53 *The American Political Science Review* 69; Seymour Martin Lipset, *Political Man: The Social Bases of Politics* (expanded edn, The Johns Hopkins University Press 1981) 27–63.

<sup>&</sup>lt;sup>31</sup> Rome Statute (n 5) art 13(b).

<sup>&</sup>lt;sup>32</sup> WB Gallie, 'Essentially Contested Concepts' (1956) 56 *Proceedings of the Aristotelian Society* 167; see also Jeremy Waldron, 'Is the Rule of Law an Essentially Contested Concept (in Florida)?' (2002) 21 *Law and Philosophy* 137.

<sup>&</sup>lt;sup>33</sup> Aristotle, *Nicomachean Ethics* (H Rackham tr, Harvard University Press 1926) Book 3, 1131a10– b15; Aristotle, *Politics* (H Rackham tr, Harvard University Press 1932) Book 3, 1131a10–b15, 12.1282b18–23. The maxim is subject to different interpretations and is contested by some scholars; no two cases are perfectly alike. See Benjamin Johnson and Richard Jordan, 'Should Like Cases be Decided Alike? A Formal Analysis of Four Theories of Justice', 21 Feb 2018, https://ssrn.com/ abstract=3127737.

<sup>&</sup>lt;sup>34</sup> Clarke (n 24); Clarke, Knottnerus and de Volder (n 24), Taku (n 24) 338.

not accept, sometimes facing criminal referrals from 'states' they do not recognise, and subject to legal determinations they contest.

Facing legitimacy challenges from within and without, the ICC finds itself in a bind, which will continue to shape and challenge its agenda. Section 2 of this article explains the ICC and its agenda, codified as the Rome Statute, as a political outcome, shaped largely by European norms and security interests of party states. Section 3 shows that politics always hangs over the head of the Court: the Prosecutor and the Court are constrained by and responsive to politics, some of which is intrinsic to the oversight and governance functions established in the Rome Statute, and other that is extrinsic to institutional rules. It also elaborates a dynamic historical account of the operation of the ICC, its legitimation crises, and its efforts to adapt, outlined above. Section 4 briefly describes some important political consequences of ICC actions in situation states, which should be taken into account in decisions of the Prosecutor and the Court, and informs strategic interaction between the Prosecutor and leaders in situation countries.

Section 5 concludes that the Court's most fundamental challenge is that it must be responsive to both its legal mandate and international political demands to survive. In applying the law of the Rome Statute, the Court has triggered claims of bias, assertions of the Court's illegitimacy, and political confrontation with and retaliatory actions by powerful party and non-party states. Yet adapting to these challenges by taking actions attentive to politics risks the ICC's legitimacy as a legal institution. While the tension between law and politics can be dampened by various changes in ICC policies, that tension and associated challenges cannot be fully resolved.

# 2. Explaining the ICC agenda and membership: European norms and the security interests of states

States that have become parties to the Rome Statute are those that accept the particular principles, rules and procedures set forth in the Statute (which embody largely European norms about laws of war), and that perceive their security is not weakened by joining the regime.

Ideas – beliefs held by people, which may be normative, ontological or epistemological – explain many foreign policies, international institutions and other political outcomes.<sup>35</sup> While it is frequently asserted that human rights are 'universal', the particulars of human rights and humanitarian law embodied in the Rome Statute are deeply rooted in western norms centred in Europe. These norms, hardened into laws of war and accompanied by western rules of procedure and practice, drove the development, form and content of the Rome Statute.<sup>36</sup> For many states and rulers, those norms have shaped

<sup>&</sup>lt;sup>35</sup> Judith Goldstein and Robert O Keohane (eds), *Ideas and Foreign Policy: Beliefs, Institutions, and Political Change* (Cornell University Press 1993).

<sup>&</sup>lt;sup>36</sup> Steven C Roach, Politicizing the International Criminal Court: The Convergence of Politics, Ethics, and Law (Rowman & Littlefield 2006); Michael J Struett, The Politics of Constructing the International Criminal Court: NGOs, Discourse, and Agency (Palgrave Macmillan 2008); see also Margaret E Keck

their perception of the national interest. Yet those norms, and the particular legal definitions, rules and procedures into which they hardened, are not shared universally. Moreover, they stand in opposition to security interests of states and rulers in much of the world. Understanding the normative roots of the Rome Statute, and the security interests of states and rulers, helps to explain both the Rome Statute agenda and limits on the scope of ICC membership.

# 2.1. Western lineage

Following the International Military Tribunals after the Second World War, slow and halting negotiations to establish a permanent international criminal court ensued through the International Law Commission, the UN General Assembly and other venues, but it was not until 1998, at a moment of western unipolarity, that the ICC was finally constituted.<sup>37</sup> The Soviet Union had fallen; Russia was not very strong; and China possessed a fraction of its current economic and military power.

In that context, it is not surprising that the Rome Statute's agenda was supported largely by the west, reflecting western geopolitical interests in maintaining a territorial status quo. Yet among western and western-influenced states, there were significant political cleavages – transatlantic and northsouth – and those cleavages influenced the Rome Statute agenda and membership.

The crimes and procedural rules defined in the Rome Statute are of European origin. Many people think of the laws and institutions addressing war crimes and genocide, and international criminal law more generally, as rooted in the Nuremburg trials at the end of the Second World War. Its foundations are much older than that.<sup>38</sup>

Probably the most important modern western work in the development of international criminal law is Hugo Grotius, *Laws of War and Peace*,<sup>39</sup> written during the Thirty Years War, which at that time was the most destructive war Europe had experienced. Grotius rested many premises on the work of Ancient Greek and Roman philosophers, as well as Thomas Aquinas<sup>40</sup> and other scholasticists,<sup>41</sup> and employed natural law reasoning, deducing central

and Kathryn Sikkink, *Activists, Beyond Borders: Advocacy Networks in International Politics* (Cornell University Press 1998); Anthea Roberts and others (eds), *Comparative International Law* (Oxford University Press 2018) Pt 6.

<sup>&</sup>lt;sup>37</sup> Richard H Steinberg, 'The Rise and Decline of a Liberal International Order' in David Sloss (ed), *Is the International Legal Order Unraveling*? (Oxford University Press 2022) 37.

 $<sup>^{38}</sup>$  For a survey of the evolution of international criminal law from ancient Greece to the late twentieth century see McCormack (n 10).

<sup>&</sup>lt;sup>39</sup> Hugo Grotius, On the Law of War and Peace (Francis W Kelsey tr, Clarendon Press 1925).

<sup>&</sup>lt;sup>40</sup> Thomas Aquinas, *Summa Theologiae* (Fathers of the English Dominican Province trs, Benziger Brothers 1947) especially Pts I–II, Question 95 'Of Human Law'.

<sup>&</sup>lt;sup>41</sup> Robert Beck, Anthony Clark Arend and Robert D Vander Lugt, 'Natural Law' in Robert J Beck, Anthony Clark Arend and Robert D Vander Lugt (eds), *International Rules: Approaches from International Law and International Relations* (Oxford University Press 1996) 34–37; Alfred Verdross

principles from those thinkers, and from the 'Old Law' (the Torah) and the 'New Law' (the New Testament).<sup>42</sup> Using 'right reason', making deductions from those western sources, Grotius developed what are foundational principles of international humanitarian law today: (i) *jus ad bellum*: law on the right to wage war – namely, the principle that the only legitimate reasons for use of force are self-defence or righting a wrong; and (ii) *jus in bello*: law in warfare, establishing a civilian-military distinction, according to which military targets in wartime are legitimate but civilian targets are not, and a principle of proportionality, that in the use of force, military action must be proportionate to the wrong being righted.

These fundamental principles, of European pedigree, are echoed in the basic contemporary instruments of international criminal law. Chapter VII of the UN Charter sets the contemporary legal bases for the use of force, providing that the only legitimate legal bases are self-defence (individual or collective) or Security Council authorisation.<sup>43</sup> The Geneva Conventions and the Additional Protocols detail contemporary law in warfare. Article 48 of Additional Protocol I requires all parties to distinguish between civilians and combatants, while Article 50 defines a 'civilian', by way of reference to the Third Geneva Convention or Article 43, as 'any person who is not' a member of an armed force, militia, or openly using arms to resist an invading force.<sup>44</sup> Protocol I requires parties to 'do everything feasible to verify that the objectives to be attacked are not civilians or civilian objects',45 and to 'take all feasible precautions in the choice of means and methods of attack, with a view to avoiding, and in any event to minimizing, incidental loss of civilian life, injury to civilians and damage to civilian objects'.<sup>46</sup> The International Criminal Tribunals for the former Yugoslavia (ICTY) and Rwanda (ICTR) each defined crimes prohibiting genocide, crimes against humanity, and war crimes, again reflecting Grotian principles and incorporating much of the Geneva Conventions' elaboration of those principles.<sup>47</sup>

and Heribert Franz Koeck, 'Natural Law: The Tradition of Universal Reason and Authority' in Ronald St J Macdonald and Douglas M Johnston (eds), *The Structure and Process of International Law: Essays in Legal Philosophy, Doctrine and Theory* (Martinus Nijhoff 1983) 17.

<sup>&</sup>lt;sup>42</sup> Hedley Bull, 'The Grotian Conception of International Society' in Herbert Butterfield and Martin Wight (eds), *Diplomatic Investigations: Essays in the Theory of International Politics* (Allen & Unwin 1966) 71; Hersch Lauterpacht, 'The Grotian Tradition in International Law' in Richard A Falk, Friedrich V Kratochwil and Saul H Mendlovitz (eds), *International Law: A Contemporary Perspective* (Westview Press 1985) 10.

 $<sup>^{43}</sup>$  Charter of the United Nations (entered into force 24 October 1945) 1 UNTS XVI (UN Charter), arts 39, 51.

<sup>&</sup>lt;sup>44</sup> Protocol Additional to the Geneva Conventions of 12 August 1949 and relating to the Protection of Victims of International Armed Conflicts (entered into force 7 December 1978) 1125 UNTS 3 (AP I), arts 48, 50. Geneva Convention (III) relative to the Treatment of Prisoners of War (entered into force 21 October 1950) 75 UNTS 135.

<sup>&</sup>lt;sup>45</sup> AP I (n 44) art 57(2)(a)(i).

<sup>&</sup>lt;sup>46</sup> ibid art 50(3).

<sup>&</sup>lt;sup>47</sup> Statute of the International Criminal Tribunal for the former Yugoslavia (ICTY) (25 May 1993, as amended on 7 July 2009) arts 3–5; Statute of the International Criminal Tribunal for Rwanda (ICTR) (8 November 1994, as amended on 16 December 2009) arts 2–4.

Similarly, the Rome Statute's prohibitions and definitions of genocide,<sup>48</sup> crimes against humanity<sup>49</sup> and war crimes<sup>50</sup> are aimed in large part at upholding the Grotian *jus in bello* principles of a civilian–military distinction and proportionality, and borrow heavily from the Geneva Conventions and Additional Protocols. Article 8 *bis* upholds Grotian *jus ad bellum* principles, contextualised to the United Nations era, defining the crime of aggression as using force for any reason not permitted by the UN Charter.<sup>51</sup>

Institutionally the Court is also western. Its rules of procedure and evidence<sup>52</sup> are based largely on those used in common law and civil law countries. Procedure is a mix of adversarial legalism, as in common law countries, and inquisitorial, as in the civil law tradition.<sup>53</sup> This western bias in procedure and evidence, and in which only certain types of domestic criminal justice process satisfy the ICC's principle of complementarity,<sup>54</sup> have been a focus of critique from Islamic law perspectives,<sup>55</sup> Asian perspectives,<sup>56</sup> and from commentators sympathetic to traditional or restorative forms of justice and reconciliation.<sup>57</sup>

<sup>50</sup> Article 8 prohibits, for example, 'grave breaches of the Geneva Conventions', including 'wilful killing' and 'torture or inhuman treatment': ibid art 8.

<sup>51</sup> Article 8 *bis* prohibits, in relevant part, 'the planning, preparation, initiation or execution ... [of] the use of armed force by a State against the sovereignty, territorial integrity or political independence of another State, or in any other manner inconsistent with the Charter of the United Nations': ibid art 8 *bis*.

<sup>52</sup> ICC RPE (n 6).

<sup>53</sup> Kai Ambos, 'International Criminal Procedure: "Adversarial", "Inquisitorial" or Mixed?' (2003) 3 International Criminal Law Review 1.

<sup>54</sup> Rome Statute (n 5) arts 17, 53.

<sup>55</sup> Mohamed Elewa Badar, 'The International Criminal Court, Islamic Legal Tradition, and the Arab World: Quo Vadis?', *ICC Forum*, https://iccforum.com/legal-traditions; Shaheen Sardar Ali and Satwant Kaur Heer, 'What is the Measure of "Universality"? Critical Reflections on "Islamic" Criminal Law and Muslim State Practice vis-à-vis the Rome Statute and the International Criminal Court' in Tallyn Gray (ed), *Islam and International Criminal Law and Justice* (Torkel Opsahl Academic EPublisher 2018) 175, 175–79.

<sup>56</sup> Motoo Noguchi, 'Criminal Justice in Asia and Japan and the International Criminal Court' (2006) 6 *International Criminal Law Review* 585; but see Amrita Kapur, 'Asian Values v. The Paper Tiger: Dismantling the Threat to Asian Values Posed by the International Criminal Court' (2013) 11 *Journal of International Criminal Justice* 1059. See also Randall Peerenboom, 'Beyond Universalism and Relativism: The Evolving Debates about "Values in Asia" (2003) 14 *Indiana International & Comparative Law Review* 1.

<sup>57</sup> Ray Nickson, 'By Recognizing Broader, Deeper, and Longer Conceptions of Justice through Complementarity, the ICC Can Transcend a Narrow, Western Approach to International Criminal Justice', *ICC Forum*, https://iccforum.com/legal-traditions#Nickson; Theresa Sophia Reinold, 'The International Criminal Court, the Global South, and the Project of Global Constitutionalism', *ICC Forum*, https://iccforum.com/legal-traditions#Reinold; Jennifer J Llewellyn, 'A Comment on the Complementary Jurisdiction of the International Criminal Court: Adding Insult to Injury in

 $<sup>^{48}</sup>$  Article 6 prohibits, for example, 'killing' or 'causing serious bodily or mental harm' to members of any 'national, ethnical, racial or religious group' 'with the intent to destroy' at least part of that group: Rome Statute (n 5) art 6.

<sup>&</sup>lt;sup>49</sup> Article 7 prohibits, for example, 'murder', 'extermination', 'enslavement', and 'deportation or forcible transfer of population' 'as part of a widespread or systematic attack against any civilian population': ibid art 7.

Finally, the model for the Rome Statute, from which much of its text is lifted, was the statute of the ICTY, drafted largely by rapporteurs appointed by the Conference on Security and Cooperation in Europe, as well as by commissions of jurists from France and Italy.<sup>58</sup> It is hard to argue that the Rome Statute is not a fundamentally European instrument.

# 2.2. Arab states, Islamic Law, and the Rome Statute: An example of ideational differences

The western principles and definitions embedded in the Rome Statute, with deep roots in Judeo-Christian thought, conflict with ideas in many other societies. Most non-western societies maintain some deeply pedigreed approaches to what is permissible in warfare, but they differ in important ways from the Rome Statute. Confucian cultures can point to Sun Tzu's Art of War, which requires humane treatment of captives,<sup>59</sup> but there are no other meaningful humanitarian prescripts in the work; moreover, it advocates plundering the enemy's territory, which is a war crime under the Rome Statute,<sup>60</sup> and the focus of the work is on maximising the prospects of military victory, not on humanitarian norms. The Manusmriti from 200 BC may be cited as an authoritative repository of Hindu law that elaborates several just war principles, stating that war should be avoided by negotiation and reconciliation; that if war becomes necessary, a soldier must never harm unarmed civilians, noncombatants, or someone who has surrendered, and use of force should be proportionate.<sup>61</sup> Neither of these works may be considered anywhere near as comprehensive or detailed as western humanitarian principles and laws. Moreover, while some of these non-western principles or laws concerning warfare bear similarities with those of the west, the rules diverge in various important respects.

By way of example, it is useful to consider how particular ideas, definitions, and prohibitions in the Rome Statute conflict with at least some interpretations of Sharia,<sup>62</sup> particularly in many Arab states in and around the Persian Gulf.<sup>63</sup> Divergent interpretations of Sharia within Arab states, where some interpretations run contrary to the Rome Statute, suggest there is polarisation

<sup>63</sup> Badar (n 55).

Transitional Contexts?' (2001) 24(2) *Dalhousie Law Journal* 192; Martha Minow, 'Do Alternative Justice Mechanisms Deserve Recognition in International Criminal Law? Truth Commissions, Amnesties, and Complementarity at the International Criminal Court' (2019) 60 *Harvard International Law Journal* 1.

<sup>&</sup>lt;sup>58</sup> William Schabas, *An Introduction to the International Criminal Court* (Cambridge University Press 2011) 11–22, 158.

<sup>&</sup>lt;sup>59</sup> Sun Tzu, 'Art of War' in T'ao Han-chang, *Sun Tzu's Art of War: The Modern Chinese Interpretation* (Yuan Shibing tr, Sterling Publishing 1990) 94–128, 109.

<sup>&</sup>lt;sup>60</sup> Rome Statute (n 5) art 8(2)(b)(xvi).

<sup>&</sup>lt;sup>61</sup> Manu (lawgiver), *The Laws of Manu* (Wendy Doniger with Brian K Smith trs, Penguin 1991). See also Nagendra Singh, *India and International Law* (S Chand & Co 1973) 4.

<sup>&</sup>lt;sup>62</sup> Kamari Maxine Clarke, Fictions of Justice: The International Criminal Court and the Challenge of Legal Pluralism in Sub-Saharan Africa (Cambridge University Press 2009) 161.

or lack of consensus within those countries on Rome Statute particulars, affecting whether certain Arab states could commit to the Statute.<sup>64</sup>

The Islamic legal tradition is characterised by a range of views and debates that vary across time and space.<sup>65</sup> Many Islamic law scholars argue that the Qur'an not only contains 'justice' as one of its core principles, but delimits war in several ways that are consistent with western and Rome Statute norms, such as permitting war to be waged only in self-defence,<sup>66</sup> distinguishing between civilians and combatants, prohibiting execution or improper treatment of prisoners of war,<sup>67</sup> and prohibiting plunder and the destruction of civilian objects. Hence, all Muslim countries have signed the Geneva Conventions, and most Muslim scholars and states hold that international humanitarian law is consonant with the spirit and ultimate objectives of Islam.<sup>68</sup>

Yet some of the rules developed in the past by classical Muslim jurists are inconsistent with international humanitarian law, and the positions of some contemporary Arab state Islamic law jurists are inconsistent with the Rome Statute's embodiment of western international humanitarian law.<sup>69</sup> The conflict between some of these interpretations of Sharia and international human rights law is reflected, for example, in the practice of Arab states to make reservations, known as the 'Sharia reservations', to human rights treaties, intended to avoid incompatibilities between the treaty provisions and Sharia rules and principles.<sup>70</sup>

A few examples illustrate conflicts between some interpretations of Sharia in Arab countries and the law of the Rome Statute. Most broadly, some Islamic law scholars, jurists, leaders, and Muslim-majority states have expressed a fundamental concern about the sources of law that would be applied in their land, that joining the ICC would usurp the Sharia's exclusive jurisdiction in those states, substituting the law of God with the law of man.<sup>71</sup> This general concern

<sup>&</sup>lt;sup>64</sup> On the inverse relationship between treaty commitment and internal national polarisation on treaty rules see, eg, Charles Perrings, Michael Hechter and Robert Mamada, 'National Polarization and International Agreements' (2021) 118(50) *Proceedings of the National Academy of Sciences* 1; Joan Esteban and Gerald Schneider, 'Polarization and Conflict: Theoretical and Empirical Issues' (2008) 45(2) *Journal of Peace Research* 131; and Kenneth A Schultz, 'Perils of Polarization for US Foreign Policy' (2017) 40(4) *Washington Quarterly* 7.

<sup>&</sup>lt;sup>65</sup> Khaled Abou El Fadl, *The Search for Beauty in Islam: A Conference of the Books* (Rowman & Littlefield 2005); Khaled Abou El Fadl, *The Great Theft: Wrestling Islam from the Extremists* (HarperOne 2005).

<sup>&</sup>lt;sup>66</sup> That claim is based largely on interpretation of two prescripts in the Qur'an: *Qur'an* (Abdullah Yusuf Ali tr, 1937 tr) 2:190 and 193, https://quranyusufali.com/2.

<sup>&</sup>lt;sup>67</sup> ibid 9:5 and 47:4.

<sup>&</sup>lt;sup>68</sup> Ahmed Al-Dawoody, 'IHL and Islam: An Overview', *Humanitarian Law & Policy*, 14 March 2017, https://blogs.icrc.org/law-and-policy/2017/03/14/ihl-islam-overview.

<sup>&</sup>lt;sup>69</sup> Badar (n 55); Markus P Beham, 'Islamic Law and International Criminal Law' in Marie-Luisa Frick and Andreas Th Müller (eds), *Islam and International Law: Engaging Self-Centrism from a Plurality of Perspectives* (Brill 2013) 347, 360.

<sup>&</sup>lt;sup>70</sup> Badar (n 55); Ali and Heer (n 55).

 $<sup>^{71}</sup>$  Siraj Khan, 'Arab and Islamic States' Practice: The Sharí'ah Clause and Its Effects on the Implementation of the Rome Statute of the International Criminal Court' in Gray (n 55) 145, 151.

reverberated in particular instances in Rome Statute negotiations, such as in arguments that the detailed procedural rules and standards employed by the ICC conflict with the comparatively abstract Islamic law approach of interpreting Sharia principles,<sup>72</sup> and in the efforts of several Arab state delegations to exclude various elements of crimes of gender violence, including those related to forced pregnancy.<sup>73</sup>

Legitimate bases for the use of force are another area of some contention. As suggested above, while there was general consensus among classical Islamic jurists that self-defence is a legitimate basis for initiating warfare, there was disagreement on whether disbelief, such as polytheism, could justify warfare,<sup>74</sup> and some contemporary Islamic extremists have interpreted Sharia to permit the use of force to expel unbelievers from Muslim lands.<sup>75</sup> This basis for use of force, of course, is not recognised by Grotian principles, the UN Charter or the Rome Statute, which could treat such action as forcible transfer of a population, a crime against humanity under Article 7(2)(d) of the Rome Statute.

Similarly, as suggested above, the Qur'an distinguishes between civilians and combatants, but 'civilian' has been a subject of dispute among Islamic jurists. Most define 'civilians' to include women, children, the elderly, clergy and hired workers, but some have also treated all 'adult' males past puberty or over 15 years of age, physically able to engage in combat, as legitimate military targets.<sup>76</sup> Taken together, this definition of 'civilian' is narrower than the protected category of 'civilians' in the Geneva Conventions and Rome Statute: a widespread or systematic murder of males over 15 years of age, physically able to engage in combat, would constitute a crime against humanity under Article 7 of the Rome Statute.

An extremist Al Qaeda interpretation broadened 'the definition of active participation [in armed conflict] to include roles that directly assist the enemy', expanding the class of legitimate targets to include those providing financial and political support to the enemy.<sup>77</sup> Al Qaeda also argued that the principle of reciprocity in warfare supersedes the principle of distinction and cited Qur'an 2:194 ('attack your attacker in like manner') to permit the reciprocal targeting of civilians.<sup>78</sup> Daish (the Islamic State of Iraq and the

<sup>75</sup> Advocates of this view sometimes refer to Sahih Muslim 1767 ('It has been narrated by 'Umar b. al-Khattib that he heard the Messenger of Allah say: I will expel the Jews and Christians from the Arabian Peninsula and will not leave any but Muslim'): Sahih Muslim (tr Abdul Hamid Siddiqui) 1767.

<sup>76</sup> Salaymeh (n 74) 142.

<sup>77</sup> Quintan Wiktorowicz and John Kaltner, 'Killing in the Name of Islam: Al-Qaeda's Justification for September 11' (2003) 10(2) *Middle East Policy* 76, 88.

<sup>&</sup>lt;sup>72</sup> ibid 152.

<sup>&</sup>lt;sup>73</sup> Steven C Roach, 'Arab States and the Role of Islam in the International Criminal Court' (2005)53 *Political Studies* 143.

<sup>&</sup>lt;sup>74</sup> Khaled Abou el Fadl, 'The Rules of Killing at War: An Inquiry into Classical Sources' (1999) 89 *Muslim World* 144, 152; Lena Salaymeh, 'Comparing Islamic and International Laws of War: Orthodoxy, "Heresy", and Secularization in the Category of Civilians' (2021) 69 *The American Journal of Comparative Law* 136, 142.

<sup>&</sup>lt;sup>78</sup> ibid 86-87; Salaymeh (n 74) 146.

Levant, ISIS) legal arguments overlapped with those of Al Qaeda on this point.<sup>79</sup> These extremist views have been sharply criticised by more mainstream Muslim jurists and public figures,<sup>80</sup> but they are part of a range of views among Islamic law scholars in the Arab world, which do not accord with western doctrine embodied in the Rome Statute.

Behaviour in accord with other interpretations of Sharia mandates may also be criminal under the Rome Statute. Corporal punishments laid out in Islamic law are contrary to modern, 'universal' human rights standards, including the prohibition of torture in the Rome Statute.<sup>81</sup> In the *Al-Hassan* case, the ICC Pre-Trial Chamber agreed with the Prosecutor's contention that applying Sharia punishments in Timbuktu could have amounted to torture, and that applying Sharia law there could prove to be an organisational policy to commit a widespread or systematic attack on the civilian population. Taken together, that particular application of Sharia law could constitute a crime against humanity under Article 7(2)(e) of the Statute.<sup>82</sup>

Hence, normative differences help to explain why all European states have joined the Rome Statute, whereas only five of the 22 Arab League states have joined.<sup>83</sup> Other explanations have been offered, such as general suspicion of western powers, or some Arab autocratic leaders' fear of prosecution,<sup>84</sup> but those explanations could be epiphenomenal of the ideational differences identified above.

#### 2.3. Norms, security interests, and ICC membership

The Rome Statute's European ideational pedigree, combined with the varying security interests of states and rulers, largely explains the composition of ICC membership. Societies in which elites have internalised European human rights norms, such as some former colonies or states that mimic a perceived western script of modernity,<sup>85</sup> shared a motivation to join the ICC. Conversely, societies with ideas about sovereignty, government, justice or

<sup>82</sup> ICC, *Prosecutor v Al Hassan Ag Abdoul Aziz Ag Mahmoud*, Public Redacted Version of 'Submissions for the Confirmation of Charges', ICC-01/12-01/18-394-Red, Pre-Trial Chamber, 9 July 2019, para 82.

<sup>83</sup> The five Arab states that are Rome Statute state parties are Coromos, Djibouti, Jordan, Palestine, and Tunisia.

<sup>84</sup> Terrence L Chapman and Stephen Chaudoin, 'Ratification Patterns and the International Criminal Court' (2013) 57 *International Studies Quarterly* 400.

<sup>85</sup> See, eg, John W Meyer and others, 'World Society and the Nation-State' (1997) 103 American Journal of Sociology 144.

<sup>&</sup>lt;sup>79</sup> Salaymeh (n 74) 147.

<sup>&</sup>lt;sup>80</sup> Murad Idris, War for Peace: Genealogies of a Violent Ideal in Western and Islamic Thought (Oxford University Press 2019).

<sup>&</sup>lt;sup>81</sup> Badar (n 55); Mohamed Ibrahim Khalil, 'Islam and the Challenges of Modernity' (2004) 5(1) *Georgetown Journal of International Affairs* 97. The prohibition of torture may also be found in the Universal Declaration of Human Rights, UNGA Res 217A (III) (10 December 1948), UN Doc A/810, art 5; International Covenant on Civil and Political Rights (entered into force 23 March 1976) 999 UNTS 171, art 7; and UN Convention against Torture and Other Cruel, Inhuman or Degrading Treatment or Punishment (entered into force 26 June 1987) 1465 UNTS 85, arts 1 and 16, and is generally considered to form part of *jus cogens.* 

human rights that diverge from those embedded in the Rome Statute have had little internal motivation to join the ICC.

Moreover, many states and rulers have security interests that run in opposition to Rome Statute rules and approaches. Some autocratic rulers need to crush internal or external opposition violently in order to retain or enhance their power or authority.<sup>86</sup> Some states, particularly great powers, have defensive military entanglements or aggressive ambitions that increase the likelihood of their engagement in armed conflict in which civilians may be harmed, thereby exposing their rulers, commanders or soldiers to the risk of Rome Statute liability greater than that of weaker states' leaders.<sup>87</sup> Conversely, as is explored further in Sections 2 and 3 below, for some rulers, ICC membership offers a tool for action against opposition militias that harm civilians and commit Rome Statute crimes.

# 2.3.1. China

Consider why China did not sign the Rome Statute and has been suspicious of the Court. Ideationally, the approach of Confucian legalism to influencing human behaviour differs markedly from western legalism<sup>88</sup> and, while contested, some argue that 'Asian values' do not prioritise human rights.<sup>89</sup> Moreover, Marxist thought – modified by Lenin, Mao, and recently Xi Jinping<sup>90</sup> – not only stands in opposition to democratic governance, at least during the transition to an egalitarian communist society,<sup>91</sup> but rationalises violence against, or mass purges of, those opposed to the socialist state,<sup>92</sup> as

<sup>88</sup> Luke T Lee and Whalen W Lai, 'The Chinese Conceptions of Law: Confucian, Legalist, and Buddhist' (1978) 29 *Hastings Law Journal* 1307; Peng He, 'The Difference of Chinese Legalism and Western Legalism' (2011) 6 *Frontiers of Law in China* 645.

<sup>89</sup> For an introduction to this debate see Chang Yau Hoon, 'Revisiting the Asian Values Argument Used by Asian Political Leaders and its Validity' (2004) 32(2) *Indonesian Quarterly* 154; but see Amartya Sen, 'Human Rights and Asian Values', speech delivered at Carnegie Council for Ethics in International Affairs, New York, 25 May 1997, https://www.carnegiecouncil.org/media/series/ morgenthau/morgenthau-lectures-1981-2006-human-rights-and-asian-values; Peerenboom (n 56).

<sup>90</sup> See, eg, Giancarlo Elia Valori, 'President Xi Jinping's Diplomacy Doctrine', *Modern Diplomacy*, 11 September 2019, https://moderndiplomacy.eu/2019/09/11/president-xi-jinpings-diplomacy-doctrine.

<sup>91</sup> See, eg, Frederick Engels, *The Civil War in France* (Zodiac 1871), https://www.marxists.org/ archive/marx/works/1871/civil-war-france; Karl Marx, 'Critique of the Gotha Programme' in *Karl Marx and Frederick Engels: Selected Works in 3 Volumes* (Progress 1970); Vladimir Lenin, 'The State and Revolution: The Marxist Theory of the State and the Tasks of the Proletariat in the Revolution' in Stepan Apresyan and Jim Riordan (eds), *Lenin: Collected Works: June-September 1917*, Vol 25 (Progress 1974) 381.

<sup>92</sup> Mao Tse-Tung, 'Problems of Strategy in China's Revolutionary War' in *Selected Works of Mao Tse-Tung, Vol 1* (Foreign Languages Press 1965) 179. See also Mao Tse-Tung, 'Classes and Class Struggle' in *Quotations from Chairman Mao Tse-Tung: 'The Little Red Book'* (Foreign Languages Press 1966), https://www.marxists.org/ebooks/mao/Quotations\_from\_Chairman\_Mao\_Tse-tung.pdf.

<sup>&</sup>lt;sup>86</sup> See Abel Escribà-Folch, 'Repression, Political Threats, and Survival under Autocracy' (2013) 34 International Political Science Review 543.

<sup>&</sup>lt;sup>87</sup> See Jason Ralph, *Defending the Society of States: Why America Opposes the International Criminal Court and Its Vision of World Society* (Oxford University Press 2007) 123. See also American Servicemembers' Protection Act, 22 U.S.C. Ch 81 sub-Ch II (2002) (US) (ASPA).

well as international violence against imperialist powers,<sup>93</sup> which would likely be criminal under the Rome Statute. Chinese attitudes towards the west are influenced by the 1839–1949 'century of humiliation'<sup>94</sup> in which China was occupied and coerced into concessions, ending with western support of Chiang Kai-Shek, in opposition to Mao Tse-Tung and the Chinese Revolution. Western refusal to recognise the People's Republic of China for another few decades further strained Sino-western relations. China's suspicion of western driven international criminal adjudication, in particular, was furthered by the Tokyo Trials of 1946, with China concluding correctly that the United States made politically motivated decisions which hampered justice, planting 'seeds of distrust' of international criminal adjudication, which have lasted to the present day in China.<sup>95</sup>

Chinese government *domestic* security behaviour is consistent with the rationales of the Chinese Communist Party (CCP) for crushing internal opposition through violent means, action that would contravene the Rome Statute. Several Rome Statute provisions limit state behaviour in non-international armed conflict,<sup>96</sup> reflecting a normative view that certain restraints on violence apply not only to international armed conflict but also to how a state treats those within its borders.<sup>97</sup> In the Rome Statute negotiations, China, like many autocracies, opposed allowing any of the defined crimes to apply in non-international armed conflict or to domestic affairs.<sup>98</sup> This stance is consistent with decades of Chinese government atrocities against its own people, which include mass killings during the Cultural Revolution,<sup>99</sup> Tibetan repression,<sup>100</sup> and the contemporary genocide in Xinjiang.<sup>101</sup>

The global revisionist ambitions of the CCP constitute an *international* security basis for its opposition to the ICC. It is noted above that the Court was founded at a unipolar moment, when the west was dominant. In that context, it is not surprising that the Rome Statute effectively supports the status quo

<sup>98</sup> Bing Bing Jia, 'China and the International Criminal Court: Current Situation' (2006) 10 *Singapore Year Book of International Law* 87; Harsh Mahaseth and Ayushi Bansal, 'Asia and the ICC: The Development of International Criminal Law in a World Changing Order' (2021) 21 *International and Comparative Law Review* 162, 171.

<sup>99</sup> Yuhua Wang, 'The Political Legacy of Violence during China's Cultural Revolution' (2021) 51 British Journal of Political Science 463.

<sup>100</sup> Enze Han and Christopher Paik, 'Dynamics of Political Resistance in Tibet: Religious Repression and Controversies of Demographic Change' (2014) 217 *The China Quarterly* 69.

<sup>101</sup> Zhu (n 95) 594.

<sup>&</sup>lt;sup>93</sup> Vladimir Lenin, 'Imperialism: The Highest Stage of Capitalism' in *Lenin's Selected Works, Vol 1* (Progress 1963) 667.

<sup>&</sup>lt;sup>94</sup> Dong Wang, 'Tracing the Contours of the Unequal Treaties in Imperial China, 1840–1911' in *China's Unequal Treaties Narrating National History* (Lexington Books 2005) 9.

<sup>&</sup>lt;sup>95</sup> Dan Zhu, 'China, the International Criminal Court, and Global Governance' (2019) 73 Australian Journal of International Affairs 585.

<sup>&</sup>lt;sup>96</sup> Rome Statue (n 5) arts 8(2)(c) and (e).

<sup>&</sup>lt;sup>97</sup> There is, however, disagreement even among western states over the extent of those restrictions. For example, most European states are parties to the Protocol Second, but the United States is not; Protocol Additional to the Geneva Conventions of 12 August 1949, and relating to the Protection of Victims of Non-International Armed Conflicts (Protocol II) (entered into force 7 December 1978) 1125 UNTS 609.

global order: it is a legal-political effort to restrain violence and armed conflict, including violence that would be a means of undermining the extant global political-economic order. Geopolitically revisionist leaders, like Xi, have demonstrated or announced their willingness to use violence to change territorial borders and to revise the global order.<sup>102</sup> Rome Statute provisions aimed at delimiting violence could constrain that behaviour – another reason why contemporary revisionist powers like China are not parties to the Rome Statute, and why they consider it a potential threat. Moreover, China continues to assert legally suspect territorial claims with bordering states or entities – including India, Japan, Taiwan, and Vietnam – and clearly illegal territorial claims over the South and East China Seas.<sup>103</sup> Use of force to assert those claims could subject Chinese leaders to ICC jurisdiction, especially if it were to become a state party.<sup>104</sup>

#### 2.3.2. Russia

Similarly, the Russian government's hostility towards the Court may be explained by its internal and international security objectives, and the ideas that define them. Domestically, setting aside a brief flirtation with democracy under Yeltsin, Russia has only a Tsarist or autocratic tradition, and has long engaged in repressive atrocities perpetrated against dissidents, with millions killed during Stalin's reign, for example. Contemporary Russian fears of being subjected to ICC jurisdiction over non-international armed conflict, particularly in the Caucasian region of Russia,<sup>105</sup> have been a specific source of its hostility towards the Court.

Internationally, Putin's revisionist ambitions, which he sees as an effort to restore Russia's rightful geopolitical position, forms another basis for its opposition to the ICC. While Tsarist Russia was influential in the early development of international humanitarian law, the 1899 Hague Convention in particular, and the USSR played a key role in establishing the Nuremberg and Tokyo trials,<sup>106</sup> contemporary Russia soured on international criminal tribunals in the 1990s, when it became clear that most ICTY prosecutions would focus on long-time Slavic allies: Serbian leaders and military personnel, Bosnian Serbs, and nationals from the Republika Srpska.<sup>107</sup> Russian scholars further questioned the legitimacy and legality of the ICTY after NATO's

<sup>105</sup> Gennady Esakov, 'International Criminal Law in Russia: Missed Crimes Waiting for a Revival' (2017) 15 Journal of International Criminal Justice 371, 377.

<sup>&</sup>lt;sup>102</sup> Xi, for example, has stated that he could use force to reunify Taiwan and China: Lily Kuo, "'All Necessary Means": Xi Jinping Reserves Right To Use Force Against Taiwan', *The Guardian*, 1 January 2019, https://www.theguardian.com/world/2019/jan/02/all-necessary-means-xi-jinping-reserves-right-to-use-force-against-taiwan.

<sup>&</sup>lt;sup>103</sup> Permanent Court of Arbitration (PCA), In the Matter of the South China Sea Arbitration before an Arbitral Tribunal Constituted under Annex VII to the 1982 United Nations Convention on the Law of the Sea between the Republic of the Philippines and the People's Republic of China, PCA 2013-19, Award, 12 July 2016.

<sup>&</sup>lt;sup>104</sup> Zhu (n 95) 594, 597; Jia (n 98).

<sup>&</sup>lt;sup>106</sup> ibid 372.

<sup>&</sup>lt;sup>107</sup> Gennady Esakov, 'International Criminal Law and Russia: From "Nuremberg" Passion to "The Hague" Prejudice' (2017) 69 *Europe-Asia Studies* 1184, 1190–91.

bombing of Kosovo in 1999.<sup>108</sup> While Yeltsin's Russia signed the Rome Statute in 1998, Putin's Russia never ratified it and became openly hostile towards the Court, incensed initially by the ICC's engagement in the situation in Georgia (South Ossetia).<sup>109</sup> This was aggravated by the ICC opening a preliminary examination of the situation in Ukraine after Russia's 2014 incursion in the East,<sup>110</sup> and finally by the ICC Prosecutor's investigation of the situation in Ukraine after Russia's 2022 aggressive invasion;<sup>111</sup> Russia responded to this, in part, by charging 92 members of Ukraine's military with crimes against humanity. One scholar has suggested that engagement in armed conflicts is now a permanent facet of Russian foreign policy, driving Russia's efforts to weaken the ICC;<sup>112</sup> another has argued that Russia was particularly concerned about ICC determinations of crimes of aggression.<sup>113</sup>

#### 2.3.3. United States

The United States' decision to not join the ICC regime is more complicated than those of the states discussed above. In contrast to those states, the US government shared with Europeans support for the principle of criminal accountability for mass atrocities, but the US approach to human rights, humanitarian law, and particular definitions and procedures elaborated in the Rome Statute differ considerably from the European approach.<sup>114</sup> Moreover, the United States has national security interests that many argue demand more limited ICC jurisdiction than that favoured by European states. The United States built a global order that champions principles of democracy, human rights and accountability for mass atrocities,<sup>115</sup> but it has been ladened with global military commitments and responsibilities to maintain that order. With more than 750 military bases in 80 countries,<sup>116</sup> and military

<sup>&</sup>lt;sup>108</sup> Larisa V Deriglazova and Olga Yu Smolenchuk, 'Prosecution for Violations of International Humanitarian Law: Russia's Position' (2021) 19 *Russia in Global Affairs* 198, 204.

<sup>&</sup>lt;sup>109</sup> Roy Allison, 'Russia Resurgent? Moscow's Campaign to "Coerce Georgia to Peace" (2008) 84 International Affairs 1145.

<sup>&</sup>lt;sup>110</sup> Bakhtiyar Tuzmukhamedov, 'Russia and the International Criminal Court: From Uncertain Engagement to Positive Disengagement' in Alexander Heinze and Viviane E Dittrich (eds), *The Past, Present and Future of the International Criminal Court* (Torkel Opsahl Academic EPublisher 2021) 733; Deriglazova and Smolenchuk (n 108).

<sup>&</sup>lt;sup>111</sup> ICC, Situation in Ukraine: Notification on Receipt of Referrals and on Initiation of Investigation, ICC-01/22-2, Pre-Trial Chamber, 7 March 2022.

<sup>&</sup>lt;sup>112</sup> See Deriglazova and Smolenchuk (n 108) 211 (describing Bely's article, *Vklyuchyonnost' Rossii v* sistemu mezhdunarodnogo ugolovnogo pravosudiya po delam o voennykh prestupleniyakh (2015), which does not seem to have an English translation).

<sup>&</sup>lt;sup>113</sup> Gennady Kuzmin and Igor Panin, 'Russia' in Claus Kreß and Stefan Barriga (eds), *The Crime of Aggression: A Commentary* (Cambridge University Press 2016) 1264.

<sup>&</sup>lt;sup>114</sup> See generally Kathryn Sikkink, 'The Power of Principled Ideas: Human Rights Policies in the United States and Western Europe' in Judith Goldstein and Robert Keohane (eds), *Ideas and Foreign Policy: Beliefs, Institutions, and Political Change* (Cornell University Press 2019) 139.

 $<sup>^{\</sup>rm 115}$  The US supported the tribunals at the end of the Second World War, as well as the ICTY, ICTR and others.

<sup>&</sup>lt;sup>116</sup> David Vine, Base Nation: How U.S. Military Bases Abroad Harm America and the World (Metropolitan Books 2015).

commitments to maintain order on every continent, US military personnel would face, and have faced, substantial risks of liability under the Rome Statute, even if illegal behaviour was inadvertent or unauthorised by superiors, whereas most ICC state parties have comparatively few or no military entanglements or risks. Hence, the biggest US objection to the Rome Statute was the extent of prosecutorial discretion. The US government wanted the ICC to take only cases referred to it by the UN Security Council, decisions over which the United States has a veto. There were other objections to Rome Statute provisions that generated substantial political concern in the United States: for example, many made the contested argument that the ICC denies fundamental US constitutional due process guarantees, such as the right to trial by jury, which may not be discarded by treaty.<sup>117</sup>

The US government also has strongly objected to a Rome Statute provision that targets leaders of Israel,<sup>118</sup> the most important US military ally in the Middle East. An overtly political provision, added at Rome, could treat certain Israeli facilitators of West Bank settlements policy as war criminals, despite the wording of that provision having no pedigree as a war crime, and voluntary settlement of occupied territory by non-state actors never before having been treated as a war crime.<sup>119</sup> Rome Statute Article 8(2)(b)(viii) elaborates as a war crime '[t]he transfer, directly or indirectly, by the Occupying Powers of parts of its own civilian population into the territory it occupies'. The Fourth Geneva Convention has similar language, but with one crucial difference: Article 49(6) states that '[t]he Occupying Power shall not deport or transfer parts of its own civilian population into the territory it occupies', but does not include 'directly or indirectly', which substantially expands the prohibited behaviour. According to the official commentary on the Fourth Geneva Convention, the sixth paragraph of Article 49 was intended to prevent a practice adopted during the Second World War, by which the Nazi German government identified people of German blood and forced them to move to annexed territory.<sup>120</sup> Lebanon and Syria proposed the 'directly or indirectly'

<sup>&</sup>lt;sup>117</sup> See, eg, Steven Voigt, 'The International Criminal Court's Antagonism Toward Our Constitution and the Need for President Bush to Articulate an Acceptable Alternative', *Renew America*, 9 September 2006, https://web.archive.org/web/20061008200124/http:/www. renewamerica.us/columns/voigt/060909; John Bolton, 'Protecting American Constitutionalism and Sovereignty from International Threats', speech delivered at the Federalist Society, Washington DC, 10 September 2018, https://fedsoc.org/events/national-security-advisor-john-rbolton-address; but see Illia B Levitine, 'Constitutional Aspects of an International Criminal Court' (1996) 9 NYU Journal of International Law and Politics 27; David M Baronoff, 'Unbalance of Powers: The International Criminal Court's Potential To Upset the Founders' Checks and Balances' (2002) 4 University of Pennsylvania Journal of Constitutional Law 800. See, generally, United States Constitution, 14th Amendment; *Reid v Covert* 354 US 1 (1957).

<sup>&</sup>lt;sup>118</sup> Bolton (n 117).

<sup>&</sup>lt;sup>119</sup> See United Nations, 'States Warn Legal Committee that New International Criminal Court Must Not Be Abused for Political Ends', press release, 22 October 1999, https://press.un.org/en/ 1999/19991022.gal3117.doc.html.

<sup>&</sup>lt;sup>120</sup> Geneva Convention (IV) relative to the Protection of Civilian Persons in Time of War (entered into force 21 October 1950) 75 UNTS 287; International Committee of the Red Cross, *Commentary of* 1957 to the Convention (IV) relative to the Protection of Civilian Persons in Time of War (1957), https://ihl-

language in order to target the Israeli government's policies of allowing settlements to be established or inhabited voluntarily. Israel was obviously displeased with the Lebanese/Syrian proposal, as was the United States, but only eight states voted against the language.<sup>121</sup> While Israeli government support of settlements in the Occupied Territories may be morally reprehensible and self-defeating policy, adding the modified language to Rome Statute Article 8(2) was overtly political, a deviation from the main project of Article 8, which generally codifies well-established war crimes. It is the only crime in the Rome Statute aimed at only one state, and it is a provision that Israel's ally, the United States, could not support. Primarily for the foregoing reasons, the United States purported to 'unsign' the Rome Statute and is not a state party.

#### 2.3.4. North-south politics

Finally, north-south politics also shaped the Rome Statute, albeit in limited ways. Two examples are illustrative. First, in the early 1990s, during the International Law Commission's drafting process to create the Draft Statute for an International Criminal Court, some countries in the global south favoured including crimes beyond those that would find their way into the Rome Statute, but those crimes were excluded: for example, illicit narcotics trafficking; recruitment, use, financing, and training of mercenaries; and wilful and severe damage to the environment. Of course, these crimes have not been considered part of international humanitarian law, which was the impetus for the Rome Statute, but it is worth noting that the excluded criminal behaviour is often perpetrated by actors from the global north, from states that opposed including them in the Statute. As discussed below, the Rome Statute has been criticised for targeting African countries; some critical analysts suggest that bias is explained partly by the exclusion of the crimes above: if they had been included in the Statute, then the ICC would be pursuing more cases against perpetrators from the north.<sup>122</sup>

Second, many find it surprising that the use of chemical and biological weapons is not illegal per se under the Rome Statute.<sup>123</sup> There was a proposal to make it illegal, but the global south was opposed. Several developing

databases.icrc.org/en/ihl-treaties/gciv-1949/article-49/commentary/1958, art 49 para 6; but see Thomas S Kuttner, 'Israel and the West Bank: Aspects of the Law of Belligerent Occupation' (1977) 7 *Israel Yearbook on Human Rights* 166. See also Eyal Benvenisti, *The International Law of Occupation* (Princeton University Press 1993) 135-44.

<sup>&</sup>lt;sup>121</sup> UN Diplomatic Conference of Plenipotentiaries on the Establishment of an International Criminal Court 9th Plenary Meeting, summary record (17 July 1998), UN Doc A/CONF.183/SR.9.

<sup>&</sup>lt;sup>122</sup> Clarke (n 24) 326.

<sup>&</sup>lt;sup>123</sup> See Dapo Akande, 'Can the ICC Prosecute for Use of Chemical Weapons in Syria?', *EJIL:Talk!*, 23 August 2013, https://www.ejiltalk.org/can-the-icc-prosecute-for-use-of-chemical-weapons-in-syria. See also United Nations, 'UN Diplomatic Conference Concludes in Rome with Decision to Establish Permanent International Criminal Court', press release, 20 July 1998, L/2889, https:// press.un.org/en/1998/19980720.l2889.html (Singapore notably abstained from the vote on the adoption of the Rome Statute and expressed dismay that chemical and biological weapons were not included).

countries took the position that because possession or use of nuclear weapons would not be illegal under the Rome Statute, then they would not agree to prohibiting the possession or use of chemical or biological weapons, regarded by some as the 'poor man's' nuclear weapons.<sup>124</sup> Hence, despite overwhelming international support for criminalising the use of such weapons, as a concession to the global south their criminalisation is excluded.

# 2.3.5. Western-dominated membership

Building an institution with European ideational lineage, and Europeaninfluenced conceptions and definitions of law, procedure and practice, was a political decision. Excluding crimes favoured by much of the global south was a political decision. Including crimes associated with non-international armed conflict; establishing broad prosecutorial discretion; not limiting jurisdiction to referrals by the UN Security Council; adding a new war crime that targets Israeli settlements – all were political decisions.

The resulting agenda has had ramifications not only for the operation of the ICC, discussed below, but for its membership. While the Statute's European lineage and influence suggest neither that all western countries have joined the regime, nor that all non-western countries reject all Rome Statute norms, it should not be surprising that the ICC's western ideational bias and European influence is mirrored in the composition of states party to the Rome Statute: almost all of Europe, and most of its former colonies in North America, South America, and Africa are state parties. In contrast, only a handful of countries east of the 30<sup>th</sup> east meridian are ICC state parties (that is, east of a vertical line on a world map, roughly running east of Finland and the Democratic Republic of the Congo (DRC)); few Arab states are state parties; and a much smaller proportion of Muslim-dominated states are parties to the Rome Statute than Christian-dominated states.

# 3. Law, politics, and adaptation

Law and politics drive ICC operations. Demands and constraints on the operation of the primary organs of the Court – the Prosecutor, Chambers, and the Registry – may be conceptualised into three nested categories. The operational law of the Rome Statute includes mandates for and constraints on the organs, but on many crucial matters the Statute offers them broad discretion.<sup>125</sup> The application and interpretation of that law are nested in the ICC's constitutional governance structure: the breadth of the organs' discretion and interpretation of discretionary provisions may be limited or shaped by the Assembly of States Parties (ASP), pursuant to the Statute's constitutional governance structure are nested in raw politics: Court organs and the

<sup>&</sup>lt;sup>124</sup> William A Schabas, An Introduction to the International Criminal Court (4th edn, Cambridge University Press 2011) 138.

<sup>&</sup>lt;sup>125</sup> See Rome Statute (n 5) arts 15, 18, 39, 42, 43, 48.

<sup>&</sup>lt;sup>126</sup> ibid art 112.

ASP face pressures extrinsic to the Statute by way of state behaviour that is independent of the Statute or its formal governance structures.<sup>127</sup> These extrinsic political pressures, including threats to exit, have had the most substantial effects on the direction of the Court.

These three drivers operate dynamically and interact in ways that tell a legal and political history of the Court. During the Court's first decade, the UN Security Council and some African leaders referred several situations to the Prosecutor, who is mandated to investigate party state self-referrals, resulting in an exclusive focus on African states in its early years. This led to claims, elaborated below, of an anti-Africa bias and assertions that the ICC is a neocolonial project, challenging the Court's legitimacy, and subjecting it to a chorus of populist demands in Africa to withdraw from the ICC.<sup>128</sup> Out of necessity, the ICC adapted. The ASP took various actions to appease African party states.<sup>129</sup> Seemingly, so did Chambers and the Prosecutor, targeting several non-party states, including Israel, initially for actions not traditionally addressed by international criminal law.

### 3.1. Legal mandates as constraints?

The agenda set by negotiators at the Rome Conference, reflecting the interests and norms of party states, directs and constrains ICC organs in many important ways. For example, the Statute confers limited subject-matter jurisdiction over four core crimes: genocide, crimes against humanity, war crimes, and the crime of aggression. No other crime may be charged. Moreover, there are relatively unambiguous terms in the Statute: 'bullets', 'bombs', and 'hospital' are nouns that lend themselves to relatively straightforward application, for example.

Yet the Statute offers the Prosecutor and judges considerable flexibility. Law has ambiguities intrinsic to language,<sup>130</sup> and international law is notoriously flexible, as diplomatic negotiators often resolve drafting disagreements by adopting ambiguous language or leaving gaps that subsequently afford international actors a range of interpretations and associated behaviour.<sup>131</sup> Moreover, the law of the Rome Statute is particularly flexible, as a result of diplomatic negotiators' decisions to invest considerable discretion in the Prosecutor and Chambers. For example, based solely on the terms of the Statute, the Prosecutor has unfettered discretion about how many resources

<sup>&</sup>lt;sup>127</sup> For a similar analytical schema see Richard H Steinberg, 'Judicial Lawmaking at the WTO: Discursive, Constitutional, and Political Constraints' (2004) 98 *The American Journal of International Law* 247.

<sup>&</sup>lt;sup>128</sup> M Cherif Bassiouni and Douglass Hansen, 'The Inevitable Practice of the Office of the Prosecutor' in Steinberg (n 24) 309; Erik Voeten, 'Populism and Backlashes against International Courts' (2020) 18 *Perspectives on Politics* 407, 417.

<sup>&</sup>lt;sup>129</sup> Christa-Gaye Kerr, 'Sovereign Immunity, the AU, and the ICC: Legitimacy Undermined' (2020)41 Michigan Journal of International Law 195, 215–19.

 $<sup>^{130}</sup>$  Aristotle, *Sophistical Refutations*, *Book 4* (Frege 1948) 210 fn 2. For an example of the problem in the legal context see *Smith v United States*, 508 US 223 (1993).

<sup>&</sup>lt;sup>131</sup> Steinberg (n 127) 258–61.

to commit to one situation under investigation versus another. When investigating a situation, the Prosecutor is directed to pursue those 'criminally responsible',<sup>132</sup> but the Prosecutor may decide precisely whom to investigate, against whom to seek an arrest warrant, and how many arrest warrants to seek. Whether a particular case is admissible is based partly on whether the alleged crimes have sufficient 'gravity'.<sup>133</sup> What does that mean? What must be the scale of atrocities to deem a situation of sufficient 'gravity'? One hundred people murdered in a particular situation? A thousand killed? A million? The Office of the Prosecutor has adopted a policy document which includes several factors that it considers<sup>134</sup> – which include the scale of the crimes, the sadism with which crimes are committed, and so on – but an unweighted list of factors is merely a framework for decision making that effectively replicates the ambiguity inherent in 'gravity'.

Subject to the Pre-Trial Chamber's oversight, the Prosecutor may also decide whether to refrain from investigating if doing so 'would not serve the interests of justice'.<sup>135</sup> For millennia, philosophers and jurists have debated what is 'justice'.<sup>136</sup> Similarly, in applying Article 8(2)(b)(iv), the Court must decide whether an attack was launched with knowledge that it would cause 'incidental loss of life ... which would be clearly excessive in relation to the concrete and direct overall military advantage anticipated' – a radically ambiguous balance.

Given the ambiguity of many of the Statute's mandates, limits and standards, what then is the next line of constraints on the Court and the Prosecutor? What stops the ICC from having a run-away Prosecutor, or the opposite: a Prosecutor who ignores horrific crimes? Or a Prosecutor who is biased in investigating rebel groups to the exclusion of government forces? Or Chambers that makes law in ways that deviate from what the Court's founders intended, or from what is fair and even-handed, or from what is corrupt or politically driven?

# 3.2. Adaptation governance politics: The Assembly of States Parties

Under the Rome Statute, the Assembly of States Parties (ASP) has several authorities that could constrain, correct or direct the organs of the Court, but those authorities have been exercised modestly. The primary mechanisms are (i) an oversight authority,<sup>137</sup> which is intended to enhance the 'efficiency and economy' of the Court, and may bring to light unjust behaviour, but which does not formally intervene in judicial matters; (ii) legislative authority, which is limited by the difficulty of meeting majority or supermajority voting

<sup>&</sup>lt;sup>132</sup> Rome Statute (n 5) art 25.

<sup>&</sup>lt;sup>133</sup> ibid art 17(1)(d).

<sup>&</sup>lt;sup>134</sup> Office of the Prosecutor of the International Criminal Court, 'Policy Paper on Case Selection and Prioritization', 15 September 2016, 12–14, https://www.icc-cpi.int/sites/default/files/ itemsDocuments/20160915\_OTP-Policy\_Case-Selection\_Eng.pdf.

<sup>&</sup>lt;sup>135</sup> Rome Statute (n 5) art 53(1)(c).

<sup>&</sup>lt;sup>136</sup> Gallie (n 32); Waldron (n 32).

<sup>&</sup>lt;sup>137</sup> Rome Statute (n 5) art 112(2)(b).

thresholds; (iii) budgetary control, which has been used to enhance efficiency, but never to influence particular cases or situations;<sup>138</sup> and (iv) election of the Prosecutor, Registrar and the judges.<sup>139</sup>

#### 3.2.1. Oversight

The oversight authority is limited by the principles of non-interference in prosecutorial activities and judicial independence.<sup>140</sup> Article 112(2)(b) of the Rome Statute provides that the ASP shall provide management oversight of the Presidency, the Prosecutor, and the Registrar regarding the administration of the Court; Article 112(4) of the Statute provides that the Assembly may establish such subsidiary bodies as may be necessary, including an independent oversight mechanism for inspection, evaluation, and investigation of the Court, in order to enhance its efficiency and economy. However, Article 42(1) provides that the Office of the Prosecutor shall act independently as a separate organ of the Court, and Article 42(2) provides that the Office shall be headed by the Prosecutor, who shall have full authority over the management and administration of the Office, including the staff, facilities, and other resources.

Oversight usually has originated with the actions of one or a few states that build a broader consensus within the ASP. As in most international organisations, the policy process is typically initiated by a powerful state or coalition of like-minded states, which slowly expands the circle of states involved in developing a proposal, until it is presented as an action item to the plenary body as a whole.<sup>141</sup> The United Kingdom, for example, launched a high-profile effort to reform the ICC at the 17th ASP meeting in 2018, suggesting that the Court had spent too much money for too few convictions, made insufficient use of performance indicators to increase its efficiency, failed to investigate allegations that had been made about the first Prosecutor, and not adequately respected the principle of complementarity.<sup>142</sup> The Foreign and Commonwealth Office followed the ASP statement by convening a subgroup of state parties at Lancaster House (London) in May 2020 to consider Court reforms, an action resented by excluded states. Court President Chile Eboe-Osuji then

<sup>141</sup> For a description of this diplomatic process in the context of international trade negotiations see Richard H Steinberg, 'In the Shadow of Law or Power? Consensus-Based Bargaining and Outcomes in the GATT/WTO' (2002) 56 *International Organization* 339.

<sup>142</sup> 'The United Kingdom of Great Britain and Northern Ireland General Debate Statement', speech delivered at the Assembly of States Parties to the Rome Statute of the International Criminal Court 17th Session, 5 December 2018, https://asp.icc-cpi.int/sites/asp/files/asp\_docs/ASP17/GD%20UK%2005-12-2018.pdf.

<sup>&</sup>lt;sup>138</sup> ibid art 112(2)(d).

<sup>139</sup> ibid arts 36, 42(4), 43(4).

<sup>&</sup>lt;sup>140</sup> ibid arts 40(1), 42(1). For discussions of the independence of the Office of the Prosecutor and its limits see Jose Alvarez, 'The Proposed Independent Oversight Mechanism for the ICC' in Steinberg (n 24) 143; Nicholas Cowdery, 'The Independent Oversight Mechanism Does Not Have Authority to Investigate and Decide Alleged Misconduct by Staff in the Office of the Prosecutor' in Steinberg, ibid 154; Max Du Plessis and Christopher Gevers, 'The Role of the Assembly of States Parties for the ICC' in Steinberg, ibid 159; Harmen van der Wilt, 'A Reasonable Request: Requiring Prosecutor Authorization Prior to Any Investigation by the Independent Oversight Mechanism' in Steinberg, ibid 178.

reinvigorated the reform effort, sending a letter to the ASP President, requesting the establishment of an independent expert assessment of Court practice, which led to the establishment of an Independent Expert Review (IER) at the ASP in December 2019. That review generated 384 recommendations for reform, covering all aspects of the Court's operations, including governance, human resources, ethics, budget, and use of performance indicators;<sup>143</sup> Chambers' working methods, efficiency, and processes and procedures; prosecutorial criteria for case selection, prioritisation, hibernation, and closure; and Registry policies on victim representation.<sup>144</sup> An ASP-approved process has been under way since then to implement many of those reforms,<sup>145</sup> and while IER-recommended reforms may subtly shift the direction of the Court, perhaps increasing efficiency and performance, the initiative has not interfered in particular matters before the Court.

Hence, formal ASP oversight has not been used to influence judicial or prosecutorial behaviour substantially on particular matters. Effective oversight may help the Court to accomplish its objectives by bringing a systematic, disciplined approach to evaluate and improve 'the efficiency and economy' of the Court, assisting the Assembly and the various organs of the Court in the effective discharge of their responsibilities. At the same time, the independence of the Prosecutor and Chambers is critical for the credibility of the Court, so the ASP and the Independent Oversight Mechanism have engaged with the organs in only limited ways, and they have never interfered in decisions to investigate particular situations or to prosecute particular individuals.

### 3.2.2. Legislative authority

The ASP is further endowed with legislative authority. Article 112(7) of the Rome Statute provides that the ASP should strive to take decisions by consensus, but that if a consensus cannot be reached, the ASP may take procedural decisions by a simple majority of state parties present, or substantive decisions by a two-thirds majority of those present (provided there is a quorum). Rome Statute amendments require adoption by two-thirds of state parties.<sup>146</sup> In practice, those majorities and supermajorities have not formed easily. The highest-profile legislative actions have included amendments to fully establish the crime of aggression and activation of the amendments. Combined with the

<sup>&</sup>lt;sup>143</sup> An earlier effort to employ performance indicators that could help the Court in developing effective and efficient practices was initiated by Switzerland: Silvia Fernández de Gurmendi, 'Remarks to the 26th Diplomatic Briefing', speech delivered at the International Criminal Court, The Hague, 9 October 2017, https://www.icc-cpi.int/sites/default/files/itemsDocuments/26db-Pres-Eng.pdf; ICC, 'Second Court's Report on the Development of Performance Indicators for the International Criminal Court', 11 November 2016, https://www.icc-cpi.int/sites/default/files/itemsDocuments/ICC-Second-Court\_report-on-indicators.pdf. Those performance indicators were never fully implemented.

<sup>&</sup>lt;sup>144</sup> ICC, 'Independent Expert Review of the International Criminal Court and the Rome Statute System: Final Report', 30 September 2020, ICC-ASP/19/16.

<sup>&</sup>lt;sup>145</sup> 'Review of the ICC – A Good Start but Still a Bumpy Ride Ahead', *Journalists for Justice*, 29 July 2022, https://jfjustice.net/review-of-the-icc-a-good-start-but-still-a-bump-ride-ahead.

<sup>&</sup>lt;sup>146</sup> Rome Statute (n 5) art 121(3).

norm of non-interference in judicial or prosecutorial decisions, the ASP has never taken a substantive or procedural decision that has interfered with a specific investigation, prosecution or judgment.

#### 3.2.3. Budgetary authority and contributions

Budgets are used commonly to control or direct organisational behaviour.<sup>147</sup> ICC operations are funded by state parties and the Court's annual budget is approved by the ASP. The Court has spent over 2 billion euros since its inception, and the ASP approved an annual budget of 155 million euros in 2022.<sup>148</sup> Various factions of state parties, often complaining of inefficiency, have limited their contributions, supported a reduced budget, or adopted a policy of zero nominal growth of the budget. However, neither the level nor the allocation of the budget has ever been used by the ASP in conjunction with a mandate to any Court organ to pursue or refrain from a particular investigation or prosecution, or to pressure Chambers to take any particular position on the law.

#### 3.2.4. Election of the Prosecutor, judges, and the Registrar

Elections are inherently political, and the ASP elects the Prosecutor, the Registrar, and the judges – the most important figures at the ICC.

The composition of the bench reflects the political nature of the elections, yet there is no evidence that elections have directly skewed judicial opinions. Nearly a third of ICC judges have held some form of diplomatic position for their country, leading some to question the relevant legal knowledge and legal analytic skills, independence, and impartiality of those former diplomats.<sup>149</sup> Lengthy and expensive campaigns are often required for judicial nominees and poorer party states are less likely than wealthier states to successfully back their nominees. Voting blocs of the state parties, and vote trading, operate to drive results that are highly politicised. In 2011, the ASP established an Advisory Committee on Nominations of Judges,<sup>150</sup> but many suggest that the Committee's assessments have lacked rigour and have had minimal impact on the voting process or on limiting political interference in the election of unqualified judges. Taken together, these practices have yielded a bench of judges who are not uniformly skilled in legal analysis, not geographically representative of the state parties, and allegations that some are not impartial, although few - other than non-prevailing parties in judicial decisions - have claimed that any particular judicial decision has been politicised.

<sup>&</sup>lt;sup>147</sup> James G March and Hebert A Simon, *Organizations* (John Wiley & Sons 1958).

<sup>&</sup>lt;sup>148</sup> 'Resolution of the Assembly of States Parties on the Proposed Programme Budget for 2022, the Working Capital Fund for 2022, the Scale of Assessment for the Apportionment of Expenses of the International Criminal Court, Financing Appropriations for 2022 and the Contingency Fund', 9 December 2021, Res ICC-ASP/20/Res.1.

<sup>&</sup>lt;sup>149</sup> Open Society Foundations, 'Raising the Bar: Improving the Nomination and Election of Judges to the International Criminal Court', 28 October 2019, 35, https://www.justiceinitiative.org/uploads/a43771ed-8c93-424f-ac83-b0317feb23b7/raising-the-bar-20191112.pdf.

<sup>&</sup>lt;sup>150</sup> ICC, 'Establishment of an Advisory Committee on Nominations of Judges of the International Criminal Court', 30 November 2011, ICC-ASP/10/36, Annex; ICC, 'Strengthening the International Criminal Court and the Assembly of States Parties', 21 December 2011, ICC-ASP/10/Res.5.

The ASP elections of the second and third Prosecutors were transparently political, but again there is little evidence of a direct effect on any particular investigation or prosecution. There have been three ICC Prosecutors. Election of the first Prosecutor, Luis Moreno Ocampo, was the least politicised. As a young lawyer, Ocampo had been deputy prosecutor in the trial of the Argentine military junta; immediately prior to his election he was the Robert F. Kennedy Visiting Professor at Harvard Law School and a board member of Transparency International. He was elected by consensus.

By the end of Ocampo's tenure, all of the Office of the Prosecutor investigations involved Africa; many of the continent's leaders were claiming that Africa was being targeted unfairly; and an African Union summit had resolved not to enforce arrest warrants issued against African leaders. There was strong pressure in some quarters for Ocampo's replacement to be African.<sup>151</sup> Many hoped that the election of Ocampo's successor would be free from political considerations and centred exclusively on electing the most qualified nominee. The ASP elected Fatou Bensouda of Gambia, who had been Ocampo's deputy and was unquestionably gualified, but while others on the short list were also qualified, Bensouda was the only short-listed African. Throughout her tenure, Bensouda argued that the ICC was not biased against Africa; she declined to open any new investigations in Africa via her proprio motu authority - except in Burundi, a tiny, weak country, where crimes had been committed too overtly to ignore - and instead opened investigations outside Africa, in Afghanistan, Bangladesh/Myanmar, Georgia, Palestine, the Philippines, Ukraine, and Venezuela.

When Bensouda's tenure ended in 2020, many African states were still complaining of an anti-Africa bias. Kenyan leaders, who had been the subject of arrest warrants for their role in 2007 post-election violence, were particularly active and vocal in their attacks on the Court. In a contested election, the ASP chose as the third Prosecutor Karim Khan. Khan is a well-respected British barrister, with substantial international criminal law knowledge and experience, who was well known at the ICC for having served as defence counsel to William Ruto, one of the Kenyan leaders who had been charged with crimes against humanity in connection with the 2007 post-election violence.

### 3.2.5. Conclusion regarding ASP governance politics

In summary, while the ASP is a political body and has influenced the organs of the Court and its direction, none of the governance mechanisms have been used to surgically direct the Prosecutor or judges with regard to any particular investigation, prosecution or decision. Some mechanisms, such as the oversight authority, have been used to modestly improve the functioning of the Court; others, such as election of the Prosecutors, have helped to assuage the political concerns of African states, without sacrificing the quality of the Prosecutors.

<sup>&</sup>lt;sup>151</sup> Lisa Clifford, 'ICC Elections: Avoid the Pitfall of Politics', *Global Policy Forum*, 7 November 2011, https://archive.globalpolicy.org/international-justice/the-international-criminal-court/general-documents-analysis-and-articles-on-the-icc/50973-icc-elections-avoid-the-pitfall-of-politics.html.

# 3.3. Adaptation and raw politics extrinsic or adjacent to the Rome Statute governance system

The most substantial political influences on the ICC have been state actions extrinsic or adjacent to the formal governance structures of the ICC. Some actions taken by party states and non-party states, respectively, are described below, followed by illustrations of ways in which those actions appear to have directly influenced the organs of the Court, triggered action or yielded inaction by the Prosecutor, or catalysed action by the ASP.

#### 3.3.1. Adapting in response to state parties' actions

State parties have pursued a range of strategies and tactics, wholly or partly extrinsic to Rome Statute procedures and institutions, to influence the operation of the ICC. For example, in response to concern about Russia's attack on Ukraine, several states provided the Office of the Prosecutor with additional funds, independent of those states' budgetary obligations, or seconded staff to the Office, to support its investigation.

Several actions by African states since 2011 illustrate strategies, outside the Rome Statute framework, that appear to have influenced the ICC. By 2011, many African heads of state and others had become concerned about the Court's direction, based on a combination of self-interest and principle. In 2009, the Pre-Trial Chamber issued an arrest warrant for Omar Al-Bashir,<sup>152</sup> a sitting head of state, despite the doctrine of immunity for sitting heads of state,<sup>153</sup> based on seemingly sound reasoning on which the Appeals Chamber would later elaborate: Rome Statute Article 27(2) provides that 'immunities or special procedural rules which may attach to the official capacity of a person, whether under national or international law, shall not bar the Court from exercising its jurisdiction over such a person'.<sup>154</sup> Later, in 2012, ICC charges were confirmed against two popular Kenyan politicians, Uhuru Kenyatta and William Ruto,<sup>155</sup> who would soon be elected President and Vice-President, respectively.

More significantly, by 2011 all of the situations under investigation by the Prosecutor were in Africa, while several glaringly criminal situations outside

<sup>154</sup> For analysis of the application of Article 27(2) and implications of a Security Council referral see Dapo Akande, 'The Legal Nature of Security Council Referrals to the ICC and its Impact on Al Bashir's Immunities' (2009) 7 *Journal of International Criminal Justice* 333.

<sup>155</sup> ICC, Prosecutor v Uhuru Muigai Kenyatta, Decision on the Confirmation of Charges pursuant to Article 61(7)(a) and (b) of the Rome Statute, ICC-01/09-02/11-382-Red, Pre-Trial Chamber II, 29 January 2012; ICC, Prosecutor v William Samoei Ruto and Joshua Arap Sang, Decision on the Confirmation of Charges pursuant to Article 61(7)(a) and (b) of the Rome Statute, ICC-01/09-01/11-373, Pre-Trial Chamber II, 4 February 2012.

<sup>&</sup>lt;sup>152</sup> ICC, Prosecutor v Omar Hassan Ahmad Al-Bashir, Warrant of Arrest for Omar Hassan Ahmad Al Bashir, ICC-02/05-01/09, Pre-Trial Chamber, 4 March 2009.

<sup>&</sup>lt;sup>153</sup> In Arrest Warrant of 11 April 2000 (Democratic Republic of the Congo v Belgium), the International Court of Justice (ICJ) reaffirmed the principle of immunity of the head of state and other high officials. The Court stated: 'In international law it is firmly established that ... certain holders of high-ranking offices, such as the head of state, head of government and minister of foreign affairs, enjoy immunities from jurisdiction in other states, both civil and criminal': ICJ, Arrest Warrant of 11 April 2000 (Democratic Republic of Congo v Belgium), Judgment [2002] ICJ Rep 3, [52].

Africa, many outside the Court's jurisdiction (such as Palestine and Syria), were not being investigated. To some, this appeared to be an anti-African bias, but of the seven African situations under investigation at that time, six had been referred to the Prosecutor in ways that left no discretion to refuse investigation. Specifically, four of the situations had been referred to the Prosecutor by the governments of the situation states themselves (Central African Republic, DRC, Kenya, Uganda),<sup>156</sup> referrals that required an investigation.<sup>157</sup> Many commentators have argued that they were referred by those governments' leaders in a self-interested effort to enlist an outside agent's actions against rebel groups operating in those countries.<sup>158</sup> Two of the seven situations had been referred to the Prosecutor by the UN Security Council (Libya and Sudan), also requiring an investigation.<sup>159</sup> Only one of the seven investigations (Côte d'Ivoire) had been undertaken pursuant to the Prosecutor's *proprio motu* authority, an investigation that required and received consideration and pre-approval by the Pre-Trial Chamber.<sup>160</sup>

The African bias claim was first asserted most prominently by leaders of three states: Kenya, Sudan, and South Africa. While the Kenyan government had referred the 2007 post-election violence situation to the Prosecutor for investigation, that government had fallen out of power, and the newly elected government was to be led by Uhuru Kenyatta and William Ruto, two targets of the ICC investigation. Kenyatta and Ruto turned the situation into a populist electoral strategy, arguing that the ICC is a neocolonial expression of European power aimed at Africa. That message was embraced by the Sudanese government, led by Omar Al-Bashir, who was the subject of an ICC arrest warrant for genocide in Darfur. The position was then adopted as a populist claim by politicians in South Africa, accepted by a majority of African governments and much of the African populace,<sup>161</sup> and expressed provocatively by the African Union Chair, Ethiopia's Prime Minister: 'The ICC is hunting Africans'.<sup>162</sup>

<sup>&</sup>lt;sup>156</sup> Pursuant to Rome Statute (n 5) arts 13(a), 14.

 $<sup>^{\</sup>rm 157}$  Unless the Prosecutor determined that there was no reasonable basis to proceed under the Rome Statue, art 53.

<sup>&</sup>lt;sup>158</sup> Schabas (n 124) 160; Paola Gaeta, 'Is the Practice of 'Self-Referrals' a Sound Start for the ICC?' (2004) 2 Journal of International Criminal Justice 949; Antonio Cassese, 'Is the ICC Still Having Teething Problems?' (2006) 4 Journal of International Criminal Justice 434, 436; Phil Clark, Distant Justice: The Impact of the International Criminal Court on African Politics (Cambridge University Press 2018).

<sup>&</sup>lt;sup>159</sup> Pursuant to Rome Statute (n 5) art 13(b).

<sup>&</sup>lt;sup>160</sup> ibid art 15.

<sup>&</sup>lt;sup>161</sup> For a thoughtful study of Kenyan public opinion on the question see Geoff Dancy and others, 'What Determines Perceptions of Bias Toward the International Criminal Court? Evidence from Kenya' (2020) 64 *Journal of Conflict Resolution* 1443.

<sup>&</sup>lt;sup>162</sup> For commentary supporting the African bias claim see Clarke (n 24); Clarke, Knottnerus and de Volder (eds) (n 24), Taku (n 24) 338. For commentary critiquing the African bias claim see Bassiouni and Hansen (n 128) 309; David Scheffer, 'Three Realities about the Africa Situation at the ICC' in Richard H Steinberg (ed), *The International Criminal Court: Contemporary Challenges and Reform Proposals* (Brill 2020) 128; Richard Dicker, 'A Court Worth Having: Defending the Integrity of the Rome Statute' in Steinberg, ibid 125; Makau W Mutua, 'Africans and the ICC: Hypocrisy, Impunity, and Perversion' in Clarke, Knottnerus and de Volder (n 24) 47.

Beginning in 2009, the African Union, which includes both party and nonparty states, began taking decisions that challenged the ICC. The Union first adopted a resolution to amend the Rome Statute to give the UN General Assembly authority to defer the proceedings of the Court and to grant heads of state immunity from prosecution during their time in office.<sup>163</sup> This was followed by another resolution directing its members to cease cooperating with the Court in executing the Bashir arrest warrant,<sup>164</sup> despite the obligation on state parties to arrest an accused for whom a warrant has been issued.<sup>165</sup> Several African state parties welcomed visits by Omar Al-Bashir, despite an arrest warrant having been issued, based on claims that the Court was biased against African states and that it was impermissible to arrest a sitting head of a non-party state, even when the situation had been referred to the Court by the UN Security Council. That legal argument was rejected by the Appeals Chamber,<sup>166</sup> and the African Union later adopted a resolution in 2018 to take its argument to the International Court of Justice.<sup>167</sup>

Even more challenging, in 2017 the African Union passed a resolution suggesting that its member states should withdraw from the Court *en masse*,<sup>168</sup> and issued a proposal for an African Criminal Court.<sup>169</sup> Gambia, South Africa, and Burundi threatened to withdraw from the ICC. Burundi did so; the Gambian government decided to remain; and South Africa reversed course only after its Supreme Court ruled that the process used by the government was unconstitutional. Exit, or the threat of exit, may be potent,<sup>170</sup> and the possibility that the campaign could gather steam, with as many as 34 African states withdrawing from the ICC, was deeply concerning.

So, how did the ICC respond? As described above, the ASP elected the second Prosecutor, Fatou Bensouda of the Gambia, an African jurist, and when her tenure ended, it elected as its third Prosecutor Karim Khan, who had been defence counsel for William Ruto in the Kenya case. The election of persons with credibility in Africa might have helped to dampen African opposition to the Court.<sup>171</sup>

<sup>165</sup> Rome Statute (n 5) art 89(1).

 $<sup>^{163}</sup>$  'Decision on the Progress Report of the Commission on the Implementation of the Assembly Decisions on the International Criminal Court (ICC)', Assembly/AU/Dec.397(XVIII), Doc EX.CL/710 (XX).

<sup>&</sup>lt;sup>164</sup> 'Decision on the Meeting of African States Parties to the Rome Statute of the International Criminal Court (ICC)', Assembly/AU/Dec.245(XIII), Doc Assembly/AU/13(XIII).

<sup>&</sup>lt;sup>166</sup> ICC, *Prosecutor v Omar Hassan Ahmad Al-Bashir*, Judgment in the Jordan Referral re Al-Bashir Appeal, ICC-02/05-01/09 OA2, Appeals Chamber, 6 May 2019.

 $<sup>^{167}</sup>$  'Decision on the International Criminal Court', Assembly/AU/Dec.672(XXX), Doc EX.CL/1068 (XXXII).

 $<sup>^{168}</sup>$  'Decision on the International Criminal Court', Assembly/AU/Dec.622(XXVIII), Doc EX.CL/ 1006(XXX).

<sup>&</sup>lt;sup>169</sup> African Union, 'Withdrawal Strategy Document', 12 January 2017, https://www.hrw.org/ sites/default/files/supporting\_resources/icc\_withdrawal\_strategy\_jan.\_2017.pdf.

<sup>&</sup>lt;sup>170</sup> Albert O Hirschman, Exit, Voice, and Loyalty: Responses to Decline in Firms, Organizations, and States (Harvard University Press 1970).

<sup>&</sup>lt;sup>171</sup> Chrispin Mwakideu, 'ICC: New Chief Prosecutor Divides Opinions in Africa', DW, 15 June 2021, https://www.dw.com/en/icc-new-chief-prosecutor-divides-opinions-in-africa/a-57892882;

What did the new Prosecutor do? Fatou Bensouda began investigations into several situations outside Africa, some in which nationals of non-state parties were the likely targets, including investigating actions of Americans in Afghanistan,<sup>172</sup> Russians in Georgia,<sup>173</sup> Russians again in Ukraine,<sup>174</sup> Burmese for crimes against the Rohingya,<sup>175</sup> and Israelis in the situation in the State of Palestine.<sup>176</sup> Bensouda also launched an investigation into the situation in the Philippines,<sup>177</sup> a state party that responded by withdrawing from the ICC (though Philippines nationals remain subject to the jurisdiction of the Court for crimes committed before withdrawal). While she used her *proprio motu* authority to pursue one new African case in a small country – the situation in Burundi,<sup>178</sup> where government figures had perpetrated flagrant and well-publicised war crimes – she did not use that authority to investigate several other situations in Africa that seemed equally grave; these included the Banyamulengue-Bembe violence in the DRC,<sup>179</sup> renewed Hema-Lendu violence there,<sup>180</sup> the killing of 500 people and displacement

<sup>177</sup> Fatou Bensouda, 'Statement of the Prosecutor, Fatou Bensouda, on Her Request to Open an Investigation of the Situation in the Philippines', 14 June 2021, https://www.icc-cpi.int/news/ statement-prosecutor-fatou-bensouda-her-request-open-investigation-situation-philippines.

Marième Soumaré, 'The African Trajectory of Karim Khan, the ICC's New Chief Prosecutor', *The Africa Report*, 3 June 2021, https://www.theafricareport.com/94521/the-african-trajectory-of-karim-khan-the-iccs-new-chief-prosecutor.

<sup>&</sup>lt;sup>172</sup> Fatou Bensouda, 'The Prosecutor of the International Criminal Court, Fatou Bensouda, Requests Judicial Authorisation to Commence an Investigation into the Situation in the Islamic Republic of Afghanistan', 20 November 2017, https://www.icc-cpi.int/news/prosecutorinternational-criminal-court-fatou-bensouda-requests-judicial-authorisation.

<sup>&</sup>lt;sup>173</sup> Fatou Bensouda, 'Statement of the Prosecutor of the International Criminal Court, Fatou Bensouda, Following Judicial Authorisation to Commence an Investigation into the Situation in Georgia', 27 January 2016, https://www.icc-cpi.int/news/statement-prosecutor-international-criminal-court-fatou-bensouda-following-judicial.

<sup>&</sup>lt;sup>174</sup> Fatou Bensouda, 'Statement of the Prosecutor, Fatou Bensouda, on the Conclusion of the Preliminary Examination in the Situation in Ukraine', 11 December 2020, https://www.icc-cpi. int/news/statement-prosecutor-fatou-bensouda-conclusion-preliminary-examination-situation-ukraine.

<sup>&</sup>lt;sup>175</sup> Fatou Bensouda, 'ICC Prosecutor, Fatou Bensouda, Requests Judicial Authorisation to Commence an Investigation into the Situation in Bangladesh/Myanmar', 4 July 2019, https://www.icc-cpi.int/news/icc-prosecutor-fatou-bensouda-requests-judicial-authorisation-commence-investigation-situation.

<sup>&</sup>lt;sup>176</sup> Fatou Bensouda, 'Statement of ICC Prosecutor, Fatou Bensouda, respecting an Investigation of the Situation in Palestine', 3 March 2021, https://www.icc-cpi.int/news/statement-icc-prosecutor-fatou-bensouda-respecting-investigation-situation-palestine.

 <sup>&</sup>lt;sup>178</sup> ICC, 'ICC Judges Authorise Opening of an Investigation regarding Burundi Situation',
9 November 2017, https://www.icc-cpi.int/news/icc-judges-authorise-opening-investigation-regarding-burundi-situation.

<sup>&</sup>lt;sup>179</sup> Rukumbuzi Delphin Ntanyoma and Helen Hintjens, 'Expressive Violence and the Slow Genocide of the Banyamulenge of South Kivu' (2021) 22 *Ethnicities* 374.

<sup>&</sup>lt;sup>180</sup> Lisa Schlein, 'UN Accuses Lendu of Mass Killings of Hema in DR Congo's Ituri Province', *Voice of America*, 28 June 2019, https://www.voanews.com/a/africa\_un-accuses-lendu-mass-killings-hema-dr-congos-ituri-province/6170782.html.

of 275,000 in Burkina Faso,<sup>181</sup> and Boko Haram's killing of 35,000 people in Nigeria since 2009.<sup>182</sup>

This turn away from Africa, towards investigations of situations involving non-state parties, required some creative jurisdictional theories. For example, Myanmar is not a state party, and the Rohingya victims are not nationals of a state party, so jurisdiction over the situation was not easily established. However, many Rohingya were forcibly displaced to Bangladesh, a state party, so the Prosecutor's investigation focuses on forced displacement, a crime that was completed in Bangladesh, conferring jurisdiction, a theory endorsed by the Pre-Trial Chamber.<sup>183</sup> The investigation of Russian actions in Ukraine is based on Ukraine, a non-party state, having conferred jurisdiction on the Court,<sup>184</sup> followed by state party referrals of the situation, triggering the investigation. Jurisdiction over the situation in Palestine entails reasoning, described below, that is creative and contested.

Not even Chambers seems to have escaped concerns about Africa. Judge Chile Eboe-Osuji, a widely respected former President of the Court, has said that many at the ICC wanted the Kenya case to 'go away'.<sup>185</sup> He has argued that '[t]he organs of the Court must take policy considerations into account' as 'failure to do so risks their legitimacy'.<sup>186</sup> Hence, on grounds of 'policy considerations', the ASP adopted Rule 134 *quater* and the Court permitted William Ruto, an accused who at the time was Kenya's Vice President, to remain in Kenya during trial hearings,<sup>187</sup> despite the clear mandate of Rome Statute Article 63(1): 'The accused shall be present at trial'.<sup>188</sup>

In 2021, the Pre-Trial Chamber affirmed Prosecutor Bensouda's request for authorisation of an investigation into the situation in the State of Palestine, another territory outside Africa, with reasoning that is controversial. In April 2012, on grounds that Palestine was not a state, Prosecutor Ocampo had rejected the Palestinian Authority's attempt to confer jurisdiction.<sup>189</sup> Subsequently, in November 2012, the Palestinian Authority went to the UN General Assembly, which voted, 138 to 9 (with 41 abstentions), to grant

<sup>&</sup>lt;sup>181</sup> Norwegian Refugee Council, 'Burkina Faso: 275,000 People Forced To Flee New Surge in Violence', 13 September 2021, https://www.nrc.no/news/2021/september/burkina-faso-275.000-displaced-since-april.

<sup>&</sup>lt;sup>182</sup> Global Centre for the Responsibility to Protect, 'Nigeria', 28 February 2023, https://www.globalr2p.org/countries/nigeria.

<sup>&</sup>lt;sup>183</sup> ICC, Situation in the People's Republic of Bangladesh/Republic of the Union of Myanmar, Decision on Requests for Leave to Submit Amicus Curiae Observations, ICC-01/19, Pre-Trial Chamber III, 14 November 2019.

<sup>&</sup>lt;sup>184</sup> Pursuant to Rome Statute (n 5) art 12(2-3).

<sup>&</sup>lt;sup>185</sup> Chile Eboe-Osuji, Speech at Stanford Law School (author's notes), 3 November 2014.<sup>186</sup> ibid.

<sup>&</sup>lt;sup>187</sup> ICC RPE (n 6) art 134 quater; ICC, Prosecutor v William Samoei Ruto and Joshua Arap Sang, ICC-01/ 09-01/11-1066, Appeals Chamber, 25 October 2013; ICC, Prosecutor v William Samoei Ruto and Joshua Arap Sang, Status Conference, ICC-01/09-01/11-T-72, Trial Chamber, 15 January 2014.

<sup>&</sup>lt;sup>188</sup> Rome Statute (n 5) art 63(1).

<sup>&</sup>lt;sup>189</sup> Office of the Prosecutor of the International Criminal Court, 'Situation in Palestine', 3 April 2012, https://www.icc-cpi.int/sites/default/files/NR/rdonlyres/C6162BBF-FEB9-4FAF-AFA9-836106D2694A/284387/SituationinPalestine030412ENG.pdf.

Palestine observer state status.<sup>190</sup> In 2015, the Palestinian Authority attempted to deposit an instrument of ratification of the Rome Statute with the UN Secretary General, who accepted it. Article 125(3) of the Rome Statute provides for this means of acceding to the Statute, but it also provides that only a state may become a state party. The President of the ICC then welcomed the 'State of Palestine' as the 123rd state party. In 2018, the Palestinian Authority referred the 'Situation in the State of Palestine' to Prosecutor Bensouda, a selfinterested move in that the clearest targets of any ensuing investigation would be the Palestinian Authority's chief nemeses - Israel and Hamas. The Prosecutor responded by launching a Preliminary Examination on 'policy grounds',<sup>191</sup> leaving open the legal question of whether the Court had jurisdiction in the matter. In 2020, Prosecutor Bensouda decided to initiate an investigation and sought a ruling from the Pre-Trial Chamber on the scope of the Court's territorial jurisdiction. That question forced the Court to consider whether Palestine is a state, because only a state could have deposited an instrument of ratification of the Statute. Forty-four amici briefs and eight government briefs<sup>192</sup> focused on three alternative possibilities.

Two of the possibilities revolved around the four 'objective' elements for determining the existence of a state, specified in the Montevideo Convention: (i) permanent population; (ii) a defined territory; (iii) a government that exercises effective control over a territory; and (iv) capacity to enter into relations with other states.<sup>193</sup> The first position was that Palestine does not have a government which controls a territory that constitutes 'Palestine'. Which government exercises control? The Palestinian Authority? Hamas? Which territory does this government control? Gaza? The West Bank? Both? 'Area C', comprising 60 per cent of West Bank territory, which is under Israeli control exclusively, under the terms of the Oslo Accords?<sup>194</sup> 'Area B', where – also under Oslo – the Palestinian Authority exercises administrative control, but shares security control with Israeli authorities? Or only 'Area A', for which the Palestinian Authority exercises exclusive administrative and police authority under Oslo? In any event, under the Oslo Accords, final borders are the subject of negotiation. If one concludes 'objectively' that Palestine is not a state, then is very hard to conclude it is a state party.

 $<sup>^{190}</sup>$  UN General Assembly, 44th Plenary Meeting Question of Palestine (29 November 2012), UN Doc A/67/PV.44.

<sup>&</sup>lt;sup>191</sup> Fatou Bensouda, 'Statement by ICC Prosecutor, Mrs Fatou Bensouda, on the Referral Submitted by Palestine', 22 May 2018, https://www.icc-cpi.int/news/statement-icc-prosecutor-mrs-fatou-bensouda-referral-submitted-palestine.

<sup>&</sup>lt;sup>192</sup> ICC Court Records, Documents related to the Situation in the State of Palestine (ICC-01/18) Amicus Curiae, https://www.icc-cpi.int/case-records?f%5B0%5D=c\_sit\_code%3A1164&f%5B1% 5D=doc\_source%3A89.

<sup>&</sup>lt;sup>193</sup> Convention on Rights and Duties of States adopted by the Seventh International Conference of American States (entered into force 26 December 1934) 165 LNTS 19 (Montevideo Convention).

<sup>&</sup>lt;sup>194</sup> Declaration of Principles on Interim Self-Government Arrangements (entered into force 13 September 1993), UN Doc A/48/486, S/26560, 11 October 1993; Israeli-Palestinian Interim Agreement on the West Bank and the Gaza Strip (entered into force 28 September 1995), UN Doc A/51/889, S/1997/357, 5 May 1997 (Interim Agreement).

A second stance is that Palestine objectively is a state under the Montevideo Convention, and is widely recognised, so it is a state party. There are plenty of precedents for interpreting the 'objective' Montevideo elements liberally. In 1960, Congo was widely recognised as a state, despite its engagement in a civil war with no effective governance over much of its territory. Similarly, even though borders were in dispute, several states have been recognised: Bosnia in 1992; Croatia in 1992; Israel in 1948. Hence, the argument goes, one is free to assert that Palestine is also a state – and so is a state party – despite having parallel competing governments in Gaza and the West Bank, and disputed borders. This liberal interpretation of the 'objective' legal criteria transforms a decision purportedly based on the Montevideo Convention into one based largely on the politics of statehood recognition, effectively consistent with a constitutive theory of statehood.<sup>195</sup>

A third stance is somewhat more nuanced and rooted in one of the most important treaties in the region, the Interim Agreement of 1995, which continues to serve as a basis for rights and responsibilities in the Occupied Territories. This treaty was at the core of the comprehensive, clear-headed, 164-page partially dissenting opinion: Judge Kovács argued that Palestine is a state, is widely recognised as a state, and is a state party, but the Court has no jurisdiction over Israelis because the Interim Agreement denies the Palestinians criminal jurisdiction over Israelis,<sup>196</sup> and a state party may not confer on the ICC criminal jurisdiction it does not have.

What did the Court do? The majority concluded it did not need to decide whether Palestine is a state! The majority laid the decision at the feet of the United Nations, arguing that Palestine is a state party by virtue of the Palestinian Authority having deposited a ratification instrument with the Secretary-General, who accepted it – and the State of Palestine encompasses the pre-1967 territory of Gaza and the West Bank, including East Jerusalem.<sup>197</sup> This argument belies the main question, as only a state could have deposited an instrument of ratification. It implies that the General Assembly's grant of observer state status decided Palestinian statehood for the Secretary General (which therefore accepted the deposit) and so for the Court. In short, a majority vote of the General Assembly made Palestine a state and a state party to the Rome Statute. Behaviourally, it is widely recognised that General Assembly way discuss issues, receive reports, and make certain recommendations,<sup>198</sup> but neither the General Assembly nor the Secretary

<sup>&</sup>lt;sup>195</sup> See generally Hersch Lauterpacht, *Recognition in International Law* (Cambridge University Press 1947); Hans Kelsen, 'Recognition in International Law: Theoretical Observations' (1941) 35 *The American Journal of International Law* 605.

<sup>&</sup>lt;sup>196</sup> Interim Agreement (n 194) art 17; ICC, Judge Péter Kovács' Partly Dissenting Opinion, ICC-01/ 18-143-Anx1, Pre-Trial Chamber, 5 February 2021.

<sup>&</sup>lt;sup>197</sup> ICC, Decision on the 'Prosecution Request pursuant to Article 19(3) for a Ruling on the Court's Territorial Jurisdiction in Palestine', ICC-01/18-143, Pre-Trial Chamber, 5 February 2021.

<sup>&</sup>lt;sup>198</sup> UN Charter (n 43) Ch 4.

General has legislative authority,<sup>199</sup> and the question before the Pre-Trial Chamber was a legal question: whether Palestine is a state. No matter. Other than international lawyers, few will see that the majority's legal analysis is wanting; the decision was politically acceptable to an overwhelming proportion of party states that diplomatically support the Palestinian cause; and it was convenient politics for an institution eager to find cases outside Africa.

# 3.3.2. Influence of non-party states

Powerful states that are not parties to the Rome Statute have also influenced ICC activities. Most importantly, the Security Council may refer (or not refer) situations of apparent criminality to the Prosecutor for investigation,<sup>200</sup> establishing ICC jurisdiction that might not otherwise exist. Of course, the permanent members of the Security Council, four out of five of which are not state parties to the Rome Statute, have weighted power in the Security Council by means of their authority to exercise a veto. In practice, such referrals have been inconsistent across situations of grave criminality. The Security Council referred the situations in Libya and Sudan (Darfur) to the Court, but has not referred some other situations of mass atrocities, such as Syria and Ukraine on account of a veto or expected veto by Russia, or the situations in Myanmar, China (Xinjiang), and China (Tibet) on account of a veto or expected veto by China. These inconsistencies seem to violate the 'like cases' maxim – treating similar cases similarly – and has ignited complaints by human rights non-governmental organisations<sup>201</sup> and from the global south<sup>202</sup> that this pattern is unfair and illegitimate.

Powerful non-party states also have leverage by offering or refusing to take actions that can assist the Court. The United States, at times, has assisted the Court and, at other moments, has attacked it. In the Court's early years, the US government was wary of the ICC, concerned that US military or civilian leaders might be subject to criminal investigation or prosecution by the Court, and enacted the American Servicemembers Protection Act (ASPA),<sup>203</sup> statutorily limiting ways in which the US government may assist or cooperate with

<sup>203</sup> ASPA (n 87).

<sup>&</sup>lt;sup>199</sup> Rosalyn Higgins, *The Development of International Law through the Political Organs of the United Nations* (Oxford University Press 1963); Stephen M Schwebel, 'The Effect of Resolutions of the U.N. General Assembly on Customary International Law' (1979) 73 *Proceedings of the Annual Meeting (American Society of International Law*) 301. The General Assembly may contribute to customary law, but in a world of sovereign states, it is hard to argue that custom binds non-consenting states.

<sup>&</sup>lt;sup>200</sup> Rome Statute (n 5) arts 13(b).

<sup>&</sup>lt;sup>201</sup> See, eg, 'UN Security Council: Address Inconsistency in ICC Referrals: Use Debate on International Court to Forge a More Principled Relationship', *Human Rights Watch*, 16 October 2012, https://www.hrw.org/news/2012/10/16/un-security-council-address-inconsistency-iccreferrals; Amnesty International, 'UN: Russian and Chinese Vetoes of Syria ICC Resolution "Callous", 22 May 2014, https://www.amnesty.org/en/latest/news/2014/05/un-russian-chinesevetoes-syria-icc-resolution-callous.

<sup>&</sup>lt;sup>202</sup> See, eg, Benson Chinedu Olugbuo, 'The African Union, the United Nations Security Council and the Politicisation of International Justice in Africa' (2014) 7 *African Journal of Legal Studies* 351; Richard Goldstone, 'The ICC and Africa' in Sharon Weill, Kim Thuy Seelinger and Kerstin Bree Carlson (eds), *The President on Trial: Prosecuting Hissène Habré* (Oxford University Press 2020) 400.

the Court and authorising certain actions that could be taken against the Court.<sup>204</sup> Understanding US concerns and preferring not to engage the United States as an active opponent of the Court, while cautiously attempting to establish the Office of the Prosecutor, the first Prosecutor, Ocampo, met with American diplomats, indicated that he trusted the US military justice system, and suggested that Americans would not be a focus of his investigations or prosecutions.<sup>205</sup> Consistent with that stance, Ocampo did not open an investigation of US behaviour in Afghanistan or in Iraq; he rejected a declaration that was intended to open an investigation into the situation in Palestine following operation 'Cast Lead'; and he focused on other cases where there were glaring atrocities in Africa, over which ICC jurisdiction was clear.

By 2008, the United States had shifted its policy towards the Court from firm opposition to a form of constructive engagement: where US interests dovetailed with those of the Court, the US government would cooperate with or support the Court's activities, to the extent consistent with the ASPA.<sup>206</sup> At the request of the National Security Council, the Department of Justice prepared a memorandum in 2010, distilling ways in which the US government may assist the Court without running afoul of the law.<sup>207</sup> These activities were determined to include furnishing certain informational assistance (including intelligence, law enforcement information, diplomatic reporting, investigative actions, and testimony) to the ICC for particular ICC cases involving foreign nationals; training ICC personnel; detailing US government employees to the ICC where assistance would be limited to particular cases involving foreign nationals; sponsoring or voting for resolutions in international fora that refer matters to the ICC or support the ICC approach to a particular matter; communicating US government views to the ICC; and encouraging foreign governments to materially assist the ICC.<sup>208</sup>

Hence, when US government policy has favoured ICC actions, the US government has assisted Court activities. For example, it provided satellite imagery to the Office of the Prosecutor, evidencing movements of armed groups in the Democratic Republic of the Congo and in Darfur; when animated, that satellite imagery shows, for example, that an armed group entered a series of villages, and shows how the villages looked before and after, demonstrating

<sup>&</sup>lt;sup>204</sup> While never used, the ASPA provides standing authorisation for use of force on the order of the President, enabling them to use special forces or other military means to rescue any American who may be detained by the Court: ibid s 7427.

<sup>&</sup>lt;sup>205</sup> David Bosco, *Rough Justice: The International Criminal Court in a World of Power Politics* (Oxford University Press 2014) 86–91; see also Moreno-Ocampo (n 2) 6 ('Our policy is never to stretch the interpretation of the norms adopted in Rome. This is the only way to build a judicial institution').

<sup>&</sup>lt;sup>206</sup> John P Cerone, 'Dynamic Equilibrium: The Evolution of US Attitudes toward International Criminal Courts and Tribunals' (2007) 18 European Journal of International Law 277; Mark D Kielsgard, Reluctant Engagement: U.S. Policy and the International Criminal Court (Martinus Nijhoff 2010) 307.

<sup>&</sup>lt;sup>207</sup> United States Department of Justice, Office of Legal Counsel, 'Memorandum for Mary DeRosa, Legal Advisor, National Security Council', 15 January 2010 (on file with the author).

<sup>&</sup>lt;sup>208</sup> ibid 22-26.

that the villages were set afire and burned to the ground.<sup>209</sup> The United States has also provided signals intelligence, intercepted cell communications and radio communications, which may be used as evidence.<sup>210</sup> It also sent 100 special forces to the Central African Republic to help to look for ICC indictee Joseph Kony. While US law prohibits its government from extraditing persons to the Court,<sup>211</sup> it nonetheless creatively assisted the Court by organising ICC indictee Bosco Ntaganda's safe passage from the US Embassy in Kigali, to which he had surrendered, to the Kigali airport to take a direct, one-way flight from Kigali to The Hague, where he was arrested on arrival.<sup>212</sup>

Conversely, there have been times when the United States ceased cooperation and instead openly attacked the Court. For example, after the second Prosecutor opened an investigation into the situation in Afghanistan, which included clear evidence of torture by US forces, and sought Pre-Trial Chamber approval of an investigation into the situation in the State of Palestine, Trump administration National Security Advisor John Bolton openly criticised the ICC, deeming it a threat to US citizens and national security;<sup>213</sup> President Trump issued an Executive Order that imposed sanctions on the Prosecutor and others in her Office, prohibiting their travel to the United States, freezing their dollar-denominated assets, and prohibiting any US person from providing material assistance to the Court under threat of criminal liability.<sup>214</sup> President Trump's successor, Joe Biden, rescinded the Trump sanctions shortly after taking office in 2021<sup>215</sup> and withdrew US forces from Afghanistan in August of that year. Almost immediately thereafter, when the Taliban resumed its rule, the third Prosecutor announced his decision to focus his office's 'investigations in Afghanistan on crimes allegedly committed by the Taliban and the Islamic State ... and to deprioritize other aspects of this investigation'.<sup>216</sup>

# 4. Political effects of the ICC in states

ICC actions and omissions have important political effects in situations under investigation. There is good evidence that, at least in its first decade of operation, ICC action advanced important objectives of those who have championed international justice. For example, statistical analyses of data across

<sup>&</sup>lt;sup>209</sup> Author interview with and demonstration by senior official in the Investigation Division, Office of the Prosecutor, ICC, The Hague (The Netherlands), 4 August 2016.

<sup>&</sup>lt;sup>210</sup> ibid.

<sup>&</sup>lt;sup>211</sup> ASPA (n 87) s 7423(d).

<sup>&</sup>lt;sup>212</sup> Jeffrey Gettleman, 'Rebel Leader in Congo Is Flown to The Hague', *The New York Times*, 22 March 2013, https://www.nytimes.com/2013/03/23/world/africa/war-crimes-suspect-bosco-ntaganda-leaves-congo-for-the-hague.html.

<sup>&</sup>lt;sup>213</sup> Bolton (n 117).

<sup>&</sup>lt;sup>214</sup> Executive Order 13928, 85 FR 36139 (2020) (US).

<sup>&</sup>lt;sup>215</sup> Executive Order 14022, 86 FR 17895 (2021) (US).

<sup>&</sup>lt;sup>216</sup> Karim AA Khan, 'Statement of the Prosecutor of the International Criminal Court, Karim A.A. Khan QC, following the Application for an Expedited Order under Article 18(2) Seeking Authorisation To Resume Investigations in the Situation in Afghanistan', 27 September 2021, https://www.icc-cpi.int/news/statement-prosecutor-international-criminal-court-karim-khan-qc-following-application.

countries have shown that, during the ICC's first decade or so, ratification of the Rome Statute was significantly associated with a reduced rate of civilian killings by party state governments, compared with the pre-ratification rate, particularly if the Prosecutor had opened an investigation or where human rights groups were operating.<sup>217</sup> Another study found that opening an ICC investigation was significantly related to prosecution of those responsible for international crimes by domestic courts in situation countries,<sup>218</sup> presumably as a result of the Rome Statute's complementarity principle, which provides that the Court cannot proceed with a case if it is being prosecuted genuinely by a state.<sup>219</sup>

At the individual country level, there is also evidence that some ICC actions have dampened violence. Joseph Kony fled from Uganda and went into hiding shortly after the ICC issued a warrant for his arrest. Ocampo's prosecutorial strategy in the DRC culminated in the arrest and removal of five powerful rebel militia and political leaders across northern and eastern theatres of conflict in Congo, and the arrest of one Congolese militia leader pursuant to an international warrant catalysed demobilisation of the militia under the arrestee's control.<sup>220</sup>

Within Israel, there is reason to believe that ICC law and activity might have influenced domestic law and practice. While the Israeli Supreme Court has never explicitly referred to the risk that Israeli nationals would be prosecuted before the ICC, decisions in recent years suggest that it is aware of that risk.<sup>221</sup> In 2006, for example, in the second targeted killings case, Chief Justice Barak's mention of international criminal tribunals' jurisprudence led the Court to decide that certain matters were justiciable,<sup>222</sup> and in the Israeli Supreme Court's 2020 Settlement Regularization Law opinion on the appropriation of Palestinian property for settlements on the West Bank, international criminal law terminology was used to deem that law to be unconstitutional.<sup>223</sup> In 2011, the Turkel Commission prepared significant military justice reforms<sup>224</sup> at precisely the time that ICC Prosecutor Ocampo was considering whether to open an investigation of the

<sup>&</sup>lt;sup>217</sup> Hyeran Jo and Beth A Simmons, 'Can the International Criminal Court Deter Atrocity?' (2016) 70 *International Organization* 443 (comparing the period 1998–2011 with a pre-ratification period); Daniel Krcmaric, *The Justice Dilemma: Leaders and Exile in an Era of Accountability* (Cornell University Press 2020) (comparing 1998–2010 with a pre-ratification period).

<sup>&</sup>lt;sup>218</sup> Geoff Dancy and Florencia Montal, 'Unintended Positive Complementarity: Why International Criminal Court Investigations May Increase Domestic Human Rights Prosecutions' (2017) 111 *American Journal of International Law* 689.

<sup>&</sup>lt;sup>219</sup> Rome Statute (n 5) art 17.

<sup>&</sup>lt;sup>220</sup> Richard H Steinberg, 'Decapitation by Arrest: International Justice and Demobilization in Congo' (unpublished paper on file with the author).

<sup>&</sup>lt;sup>221</sup> David Kretzmer and Yaël Ronen, *The Occupation of Justice: The Supreme Court of Israel and the Occupied Territories* (Oxford University Press 2021) 506–07.

<sup>&</sup>lt;sup>222</sup> HCJ 769/02 Public Committee Against Torture in Israel and Palestinian Society for the Protection of Human Rights and the Environment v Israel and Others ILDC 597 (IL 2006) [2006] para 53.

<sup>&</sup>lt;sup>223</sup> HCJ 1308/17 Silwad Municipality et al v The Knesset et al (9 June 2020), paras 40–60.

<sup>&</sup>lt;sup>224</sup> Turkel Commission, 'The Public Commission to Examine the Maritime Incident of 31 May 2010: Second Report', February 2013, Part 1, https://www.gov.il/BlobFolder/generalpage/downloads\_eng1/en/ENG\_turkel\_eng\_b1-474.pdf; Part 2, https://www.gov.il/BlobFolder/generalpage/downloads\_eng1/en/ENG\_turkel\_eng\_b475-941.pdf.

situation in the region, pursuant to a declaration of jurisdiction by the Palestinian Authority, and how to proceed on the referral by Turkey concerning the maritime incident of 2010. As the Turkel reforms were being adopted, Ocampo formally rejected the Palestinian declaration;<sup>225</sup> shortly thereafter his successor formally ended the investigation of the maritime incident of 2010.<sup>226</sup>

In contrast, some consequences of ICC action have been normatively problematic. For example, whenever a state party government has referred a case to the Prosecutor for investigation, it has served its own self-interest: targeting the behaviour of opposition militias or key rebel leaders, facilitating the Prosecutor's access to evidence against the leaders of opposing militias, and not cooperating with the Prosecutor's efforts to obtain evidence of crimes perpetrated by the referring government or allied militia leaders. In the situations in the DRC, for example, ICC arrest warrants were not sought for President Joseph Kabila or other Congolese leaders, who presided over an army that had engaged in war crimes; and ICC indictee Bosco Ntaganda, 'The Terminator', was absorbed into the Congolese army as a general, rather than being arrested by the government, as required by the Rome Statute. Similarly, arrest warrants were never sought for Ugandan President Yoweri Museveni and his cadre, despite their leadership of an army that had perpetrated atrocities. The horizontal inequity of pursuing leaders of opposition militias that perpetrated crimes, but not government leaders of armies that perpetrated crimes, might be seen as contradicting the 'like cases' maxim.

# 5. Conclusion: Squaring the circle of law and politics

The ICC can be understood only as a legal institution embedded in politics. Its agenda, codified as the Rome Statute, is a political outcome, strongly influenced by European ideas, and reflecting the normative and material interests of party states. The institution is responsive to international political pressure, particularly from state parties; and the ICC and its actions have political consequences in situation countries.

As a legal institution, however, the ICC risks losing its legitimacy if it is perceived as a political animal.<sup>227</sup> Hence, the organs of the Court must always act under the colour of law. Statements of successive Prosecutors that they will not let politics influence their decisions, quoted at the beginning of this article, are consistent with maintaining the Court's legitimacy as a legal institution. In so far as outcomes are as expected, party states have an interest in letting the Court do its work, an agenda to which they signed, without interference. Accordingly, most state parties, particularly those in Europe, have shared a norm of not pressuring the Prosecutor or Chambers to handle any particular

 <sup>&</sup>lt;sup>225</sup> The Office of the Prosecutor of the International Criminal Court, 'Situation in Palestine',
3 April 2012, https://www.icc-cpi.int/sites/default/files/NR/rdonlyres/C6162BBF-FEB9-4FAF-AFA9-836106D2694A/284387/SituationinPalestine030412ENG.pdf.

<sup>&</sup>lt;sup>226</sup> Bensouda (n 3).

<sup>&</sup>lt;sup>227</sup> Max Weber, 'The Three Types of Legitimate Rule' (Hans Gerth tr, 1958) 4(1) Berkeley Publications in Society and Institutions 1; Reinhard Bendix, Max Weber: An Intellectual Portrait (University of California Press 1977) 294.

investigation or prosecution in any particular way, expressed in the Statute as respecting the independence of the Prosecutor and Chambers.

Metaphorically, the state parties are the principal and the ICC organs are its agents for ending impunity for atrocity crimes, but there has been considerable agent slack:<sup>228</sup> politics leads the ICC on a loose leash – a leash that is sometimes vanked back to correct the Court's direction. This enables the ICC to act as a legal institution, investigating and prosecuting crimes, when raw politics might not otherwise catalyse or permit action. Moreover, as an international legal instrument, the Rome Statute may affect domestic law in party states that give direct effect to or transpose elements of it into domestic codes or court decisions. Even in some non-party states, like Israel, some international criminal law has been incorporated into domestic law via court decisions or domestic codes, perhaps because of the threat of ICC investigations or of cascading global norms.<sup>229</sup> The resulting body of domestic law and norms, combined with the ever-present threat of ICC criminal prosecution, seems to have marginally dampened civilian killings by state party governments, at least during the Court's first decade of operation, especially where human rights groups could use that law to challenge the government in court or in the media.<sup>230</sup> Being a legal institution matters.

Nonetheless, as a legal institution embedded in politics, the ICC recurrently faces legitimacy challenges – not only when it responds to politics, but also when it follows the law. It is squeezed at both ends. Legitimacy is fundamentally a subjective and normative concept, existing in the beliefs of individuals about the rightfulness of a ruler or an institution; those beliefs may be enhanced or diminished by either normative or material appeal.<sup>231</sup>

Many of the African governments, leaders, and rebels who began to challenge the ICC's legitimacy in the 2010s had both material and principled reasons for doing so, even though the Prosecutors were following their legal mandates. The ICC indictments of African leaders such as Bashir, Kenyatta, and Ruto gave them and their followers obvious material reasons for declaring the Court's actions illegitimate. Similarly, ICC-accused rebels in places like the DRC and Uganda had material reasons for claiming ICC illegitimacy as they became prosecutorial targets. However, those rebels also had principled bases for questioning the ICC's legitimacy, arguing that government leaders that referred their countries' situations to the ICC Prosecutor were not seriously investigated or charged with crimes, despite evidence that they knew their militaries were engaged in a pattern of committing crimes that they did not try to stop. However, the broadest legitimacy challenge from Africa has been the perception and argument by African state leaders of horizontal inequity, rooted in the application of the

<sup>&</sup>lt;sup>228</sup> See generally Roland Vaubel, 'Principal-Agent Problems in International Organizations' (2006) 1 *The Review of International Organizations* 125; Karen J Alter, 'Agents or Trustees? International Courts in Their Political Context' (2008) 14 *European Journal of International Relations* 33.

<sup>&</sup>lt;sup>229</sup> Kathryn Sikkink, *The Justice Cascade: How Human Rights Prosecutions Are Changing World Politics* (WW Norton & Company 2011).

<sup>&</sup>lt;sup>230</sup> Jo and Simmons (n 217); Krcmaric (n 217).

<sup>&</sup>lt;sup>231</sup> Hurd (n 1); Allen Buchanan, Justice, Legitimacy, and Self-Determination: Moral Foundations of International Law (Oxford University Press 2003).

Rome Statute's jurisdictional triggers to party states with vastly different social conditions, skewing the geographic distribution of investigations among state parties. In parts of Africa, rule of law is weak, law of war norms are weak, and state failure abounds, leaving large swathes of territory virtually anarchic and subject to atrocities perpetrated by competing militias and armies. These conditions do not exist in most other state parties, certainly not in Europe.<sup>232</sup>

At the same time, it is difficult to establish jurisdiction over situations with crimes perpetrated by nationals of non-party states, unless committed on the territory of a state party. A non-party state may confer jurisdiction on the Court for crimes committed on its territory, triggering a preliminary examination, but that can move to the investigation phase only following a state party referral, or the Prosecutor's exercise of their *proprio motu* authority, with Pre-Trial Chamber approval – a cumbersome process that has been used only rarely. The UN Security Council may also refer a situation to the Prosecutor, conferring jurisdiction and automatically triggering an investigation, but Security Council referrals have been rare because of the veto power of the permanent members, which are engaged in deepening geopolitical competition with each other. Glaringly, the UN Security Council has failed to refer the China (Xinjing), Myanmar (Rohingya), Syria or Ukraine situations to the ICC.

The resulting geographic focus of early ICC cases in Africa set off a legitimation crisis,<sup>233</sup> rooted partly in a material appeal to victimhood, and partly in principled appeals based on assertions of neocolonialism and arguments rooted in violation of the 'like cases' maxim.<sup>234</sup> ICC investigations into situations in African states exclusively, all but one of which were mandated by the Statute, while lacking authority to launch investigations in places like Syria and Palestine, became seen by many Africans as horizontally unjust, triggering a loss of confidence in the institution across much of Africa.

Some politically driven, horizontally inequitable actions that have been taken less transparently bear less salience, or do not appear to challenge the interests of any state parties, have not caused much of a stir. In some cases, those political actions seem to have enhanced the Court's legitimacy, at least for particular audiences. For example, the election of a Prosecutor from Gambia, Fatou Bensouda, seems to have been influenced by the politics of appeasing certain African states that were complaining of bias, with some threatening to exit from the ICC; that political action is likely to have enhanced the institution's legitimacy in Africa, without damaging ICC legitimacy elsewhere, as a result of Bensouda's strong qualification for the job on the merits. Similarly, the Court's decision on 'policy grounds' to permit hearings concerning Kenyan indictees in absentia, despite unambiguous Rome Statute language requiring their presence, is likely to have marginally dampened populist Kenyan complaints about the ICC, without engendering complaints.

The African legitimation crisis has been dampened most effectively, but not fully resolved, by launching investigations of several non-African situations,

<sup>&</sup>lt;sup>232</sup> See Lipset citations at n 30.

<sup>&</sup>lt;sup>233</sup> Jürgen Habermas, Legitimation Crisis (Thomas McCarthy tr, Polity Press 1988).

<sup>&</sup>lt;sup>234</sup> See Aristotle (n 33).

including investigating nationals of non-party states – for example, Afghanistan (partly targeting U.S. nationals), Georgia and Ukraine (Russians), Bangladesh (Myanmar), and Palestine (Israel). These often have been supported by creative jurisdictional theories, almost always endorsed by Chambers.

That geographic enlargement of investigations is simultaneously politically expedient, in so far as it dampens African assertions of horizontal inequity, and is consistent with the ICC mission of ending impunity for crimes defined in the Rome Statute. However, to the extent that those investigations target nationals of states that are not parties to the ICC regime, they exacerbate tension with and further delegitimate the ICC in the eyes of powerful elements in those states. For material reasons, targeted leaders and their supporters are sure to be repelled by such actions. On principle, it should not be surprising that non-party states find it unfair and unjust to have their nationals subject to the jurisdiction of a legal regime their country opted not to join, in some cases because the Rome Statute criminalises behaviour that their country does not consider criminal, follows procedures they consider unfair, or rests on referrals from 'states' they do not recognise.<sup>235</sup> These ICC investigations recapitulate ideational and interest-based disagreements in the Rome Statute negotiations, cementing the refusal of non-party states to join the Court, increasing diplomatic tension between those states and the ICC, and risking overt and covert actions against the Court. For those states, ICC investigations and arrest warrants lack legitimacy. We live in a world that is increasingly bifurcated by states that support and those that oppose the Court.

The Court's biggest contemporary political challenges are likely to continue to be from African states and powerful non-party states. Powerful elements in both categories of states – some of the world's most powerful states and some of the weakest – now question the legitimacy of the ICC. African party state governments continue to endure popular legitimacy concerns about Court bias and a perceived European neocolonial agenda – yet crimes continue to be perpetrated in Africa, promising further investigations there, as the ICC follows its legal mandates. New cases are far less likely to emerge in western Europe or developed countries, where rule of law, democracy, and observance of human rights are more likely to prevail. Non-party states – the United States, Russia, and China, for example – can be expected to challenge the Court, if they or their allies are subjected to investigation or prosecution, and to pressure their allies to reduce cooperation with the Court.<sup>236</sup>

<sup>&</sup>lt;sup>235</sup> Bolton (n 117).

<sup>&</sup>lt;sup>236</sup> As reaction to the Russian atrocities in Ukraine has shown, even when extraordinarily grave crimes are perpetrated on a large scale, if the perpetrators are nationals of powerful states, those states, their allies, and states wishing to remain non-aligned are unlikely to condemn the behaviour at issue. In October 2022, the General Assembly voted to demand Russia to reverse course in Ukraine by a vote of 143:5, with 35 abstentions. The majority of those abstaining were African nations, alongside China and India. Those states voting against the resolution or abstaining represent about half the world's population: UN, 'Ukraine: UN General Assembly Demands Russia Reverse Course on "Attempted Illegal Annexation", UN News, 12 October 2022, https:// news.un.org/en/story/2022/10/1129492.

More broadly, if world order continues to deteriorate – with revisionist powers or their proxies challenging territorial borders, perpetrating atrocities, or triggering existential wars – then the logic of the appropriateness of fighting within the confines of humanitarian law will come under increased pressure, the priorities of influential states may change, the ICC will face greater political challenges, and the antinomy between law and politics at the ICC may become perilously exposed.

ICC organs will no doubt continue trying to manage the contradictory logics of law and politics. Management of that tension might be enhanced by the Court building the capacity of the Office of the Prosecutor to analyse international and domestic politics so that it can operate more deftly in the international political environment. The Court could also help to shift the task of accountability towards national courts and regional organisations, which may enjoy more legitimacy than action in The Hague, by more aggressively pursuing a policy of positive complementarity, offering greater support to states with direct jurisdiction over a situation, states asserting universal jurisdiction, and regional organisations that are pursuing accountability.<sup>237</sup> Or the Prosecutor could raise the bar for investigations and prosecutions, pursuing only the gravest crimes that are being perpetrated on the largest scale;<sup>238</sup> these are the cases that would most glaringly burnish the ideals for which the Court stands, so they should enjoy the greatest legitimacy.<sup>239</sup> Such policy changes may modestly reduce antagonism towards the Court, but nothing can square the circle of law and politics at the ICC.

**Acknowledgements.** An earlier version of this article was prepared for the Bob and Sally Shafton Memorial Lecture, delivered at the Hebrew University of Jerusalem in March 2022. The author thanks Einat Albin, Shamila Batohi, Fatou Bensouda, Richard Dicker, Danny Evron, David Kretzmer, Jill Levenfeld, Kate Mackintosh, Luis Moreno Ocampo, Ayala Procaccia, Yaël Ronen, and Yuval Shany for insightful suggestions.

Financial statement. Not applicable.

Competing interests. The author declares none.

 $^{239}$  The Prosecutor is not obligated under the Statute to investigate all situations in which crimes are perpetrated – only those with 'sufficient gravity' that are 'the most serious crimes of concern to the international community as a whole', and that meet the other standards of jurisdiction and admissibility: Rome Statute (n 5) art 17(1)(d), art 5.

Cite this article: Richard H Steinberg, 'Politics and Justice at the International Criminal Court' (2024) 57 Israel Law Review 308–350, https://doi.org/10.1017/S0021223724000049

<sup>&</sup>lt;sup>237</sup> William Burke-White, 'Proactive Complementarity: The ICC and National Courts in the Rome System of International Justice' (2008) 49 *Harvard International Law Journal* 53; Bing Bing Jia, 'Reflections on the ICC's Relationship with the United Nations and Regional Courts' in Steinberg (n 162) 358.

<sup>&</sup>lt;sup>238</sup> For a similar view see Todd Buchwald, 'Part I: What Kinds of Situations and Cases Should the ICC Pursue? The Independent Expert Review of the ICC and the Question of Aperture', *Just Security*, 30 November 2020, https://www.justsecurity.org/73530/part-i-what-kinds-of-situations-and-cases-should-the-icc-pursue-the-independent-expert-review-of-the-icc-and-the-question-of-aperture;

but see Margaret M deGuzman, 'Gravity and the Legitimacy of the International Criminal Court' (2009) 32 Fordham International Law Journal 1400.